

Review **Urban Wetlands: A Review on Ecological and Cultural Values**

Somayeh Alikhani ^{1,2,3,*}, Petri Nummi ¹, and Anne Ojala ⁴

- ¹ Department of Forest Science, University of Helsinki, 00790 Helsinki, Finland; petri.nummi@helsinki.fi
- ² Helsinki Institute of Sustainability Science (HELSUS), 00014 Helsinki, Finland
- ³ Helsinki Institute of Urban and Regional Studies (Urbaria), 00100 Helsinki, Finland
- ⁴ Natural Resources Institute Finland (LUKE), Latokartanonkaari 9, 00790 Helsinki, Finland; anne.ojala@luke.fi
- * Correspondence: somayeh.alikhani@helsinki.fi

Abstract: Wetlands are a critical part of natural environments that offer a wide range of ecosystem services. In urban areas, wetlands contribute to the livability of cities through improving the water quality, carbon sequestration, providing habitats for wildlife species, reducing the effects of urban heat islands, and creating recreation opportunities. However, maintaining wetlands in urban areas faces many challenges, such as the reduction of hydrological functions, changed water regimes due to barriers, contamination by wastewater, habitat loss due to land-use change, and loss of biodiversity due to the entry of alien species. In this article, we review the theoretical background of wetlands in urban areas through the existing studies in the literature. We provide knowledge on urban wetlands and highlight the benefits of these wetlands in urban areas. These benefits include *sustainability, biodiversity, urban heat islands, social perception,* and *recreation* values. We also summarize the objectives, methodologies, and findings of the reviewed articles. Our review study addresses the research gaps by performing a rigorous analysis to identify significant open research challenges, showing the path toward future research in the field. We further discuss and highlight the role of policymakers and stakeholders in preserving wetlands and finally present our conclusions.

Keywords: wetland; urban wetland; green-blue infrastructure; sustainability; ecosystem services

1. Introduction

According to the United Nations, 55% of the world's population currently lives in urban areas. This number is also projected to grow to 68% by 2050 [1]. This growing tendency of urbanization, both in terms of area and density, is affecting the natural infrastructure and disrupting its natural process of sustainability [2]. It also severely affects the health and ecological functions of the urban environment, leading to vegetation degradation, water pollution, and biodiversity loss [3]. Moreover, urban areas, in most cases, have less vegetation and water compared with the surrounding areas, and existing green and blue infrastructure is often threatened by increased population density [4].

Preserving and creating these infrastructures in the urban context can also be used to control the adverse effects of climate change. Therefore, there is a need to study the impact of urban natural infrastructures, such as green-blue infrastructure (GBI), on the climate of urban areas. Using GBI, this impact of climate change can be adopted in urban areas. In cities, GBI enables water treatment, reduction of urban runoff, and the provision of psychological and social ecosystem services. For example, water ecosystems, such as ponds and other urban wetlands, which are two elements from GBI, contribute to hydrological cycles in the cities [5].

Wetlands are considered as one of the excellent natural resources in urban areas. Wetlands are one of the most important green-blue infrastructure components with their wide range of services. The optimal use of wetlands increases the social and environmental sustainability of urban areas [6]. In particular, wetlands situated in urban areas are a fundamental element of urban ecosystems. Urban wetlands provide various ecosystem services



Citation: Alikhani, S.; Nummi, P.; Ojala, A. Urban Wetlands: A Review on Ecological and Cultural Values. *Water* 2021, *13*, 3301. https:// doi.org/10.3390/w13223301

Academic Editors: Anne W. M. Ng and Nitin Muttil

Received: 15 October 2021 Accepted: 17 November 2021 Published: 22 November 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).



and vital suppliers to the human communities [7]. These include coastal area conservation [8], water quality rectification [9,10], reducing air pollution, carbon sequestration [11], and recreation and leisure [12].

The other positive effects of blue infrastructure, such as wetlands and ponds in the urban area include controlling the effect of urban heat islands (UHI). This effect is mainly due to changes in the surface properties that lead to a greater absorption of solar radiation, reduced convection cooling, and reduced water evaporation [2]. In addition, it is known that urban wetlands show their own microclimate and are usually cooler than the surrounding areas [4,13]. Thus, they help to improve the quality of life and the environment, which leads to sustainable urban development [14].

Wetlands also play an essential role in the protection of global biodiversity. Accordingly, wetlands are known as hot spots of biodiversity [15]. However, they are the most endangered ecosystems on the globe [16] since they are ecologically sensitive yet compatible systems [17]. This characteristic has caused wetlands to offer great diversity due to their origin, geographical location, water and chemical regime, dominant species, and soil and sediment features [18].

Many cities consider the conservation and restoration of urban wetlands as a strategy in urban planning that can make cities more resistant to climate change [19]. However, while wetlands play an essential role in cities and offer various services, these services are drastically under pressure due to rapid urban expansion [20]. In fact, the urbanization and the development of cities have presented wetlands with many challenges, such as (i) direct habitat loss due to land reclamation and dredging, (ii) changed water regime by barriers, (iii) contamination by wastewater, garbage, and pesticides, and (iv) biodiversity loss due to the introduction of alien species [6].

Therefore, wetland preservation has been seriously threatened by the surrounding urban development and expansion processes. It is necessary to preserve wetlands in cities to helps reduce climate change impacts. Therefore, the need to study wetlands and their effects on urban areas and their inhabitants is required.

This article aims to investigate the role of wetlands in urban areas by answering how urban wetlands contribute to the values in urban environments. The considered values in this article include *sustainability, biodiversity, urban heat islands, social perception,* and *recreation*. Each of these values benefits urban environments by reducing the impacts of climate change, enhancing the sustainability and livability of the cities, and easing accessibility of nature and water resources in urban environments, which offers diverse recreational benefits.

1.1. Motivation and Contributions

Recently, wetlands as part of the GBI, are mostly considered as nature-based solutions. They can provide many services of significant social, economic, and environmental value to human well-being [21]. At the same time, wetlands are known as ecologically sensitive systems. This knowledge clarifies why much attention has been paid in recent years to formulating and implementing sustainable management strategies for wetlands [22]. Concern about wetlands could connect ecology and society through science, partnerships, and ethics. This important step helps to realize a more integrated and interdisciplinary approach to environmental research [23].

Due to the increasing growth of cities, aquatic environments, such as lakes, wetlands, and ponds have been drawn from the outskirts of cities into the urban texture [24]. Fortunately, using this opportunity helps us meet the needs of fast-growing urban areas and adapt to climate change. Urban wetlands interact (Figure 1) with various parameters, including adverse effects of climate change, population growth and density, urban development, urbanization, social perception, sustainability, and help by improving health and well-being, bringing biodiversity to the city, and controlling the urban heat island effect.

This article reviews and analyzes the current research perspective on wetlands and their role in the urban environment. We focus specifically on how wetlands affect the urban environment by controlling the impact of the urban heat island, providing biodiversity, recreational opportunities, increasing urban sustainability, and investigating public perception. Previous studies on wetlands have focused on the benefits, economic value calculation, and ecosystem services. However, to the best of our knowledge, few review articles focus on urban wetlands to highlight their role as a solution in adapting to climate change.

In this article, we first contribute by providing a background on constructed wetlands and compare them with natural wetlands. Then, we explore the literature on urban wetlands and conduct a review about the relation between the urban wetlands and the fundamental values offered by these wetlands, including *sustainability*, *biodiversity*, *urban heat islands*, *social perception*, and *recreation* values. We also summarize the objectives, applied methodologies, and findings of the reviewed articles in five tables.

In addition, we summarize the critical research gaps addressed in the reviewed articles. Indeed, our review study addresses the gaps by performing a rigorous gap analysis to identify significant open research challenges and, hence, show the path toward future research directions in the field. Finally, we discuss and highlight the role of policymakers and stakeholders and conclude our article.



Figure 1. Ecological and cultural values offered by urban wetlands.

1.2. Methodology

We conducted a comprehensive interdisciplinary scientific literature review on the impact of wetlands on urban climate and human comfort. The databases of: ISI WEB OF KNOWLEDGE (ISI), SCIENCE DIRECT (SCD), WILEY, SPRINGER (SPR), and MDPI were searched. A search by title and abstract was performed with the following keywords: "Urban wetland", "nature-based solutions in urban areas", "Wetland as climate resilience", "Wetland as green-blue infrastructure", "Wetland ecosystem services", and "Wetland and urban sustainability". We limited the search to articles published between 2000 and 2021 to discover challenges. Table 1 shows the summary of our search from different databases.

Table 1. The keywords, numbers of articles, and the names of journal databases.

Keyword	ISI	SCD	WILEY	SPR	MDPI
Urban wetland	5	9	2	9	4
Nature-based solutions in urban areas	2	3	1	0	1
Wetland as climate resilience	4	6	3	1	1
Wetland as green-blue infrastructure	6	3	0	0	1
Wetland ecosystem services	8	6	5	4	3
Wetland and urban sustainability	8	8	1	2	4
Urban development and wetland	8	7	2	2	2
Urban wetlands and recreational use	4	6	0	2	1
Social perception of wetland	5	4	1	3	2
Total number	50	52	15	21	19

It is worth mentioning that, when searching the databases as presented in Table 1, we chose articles that studied urban wetlands from different cities in different countries. To this end, in our paper, we review a total of 48 urban wetlands located in 23 countries covering all of the continents. The locations of these urban wetlands are shown on the world google map (https://www.google.com/maps/, accessed on 15 September 2021) in Figure 2. The legend of the figure presents the name of the countries and the number of wetlands. The selection of these wetlands, each having exclusive characteristics providing valuable services in urban environments, provides a deep understanding of these wetlands' ecological and cultural values. As a result, these selections help to achieve the objectives of our review article to make a rigorous analysis and form solid conclusions.



Figure 2. The locations and number of reviewed urban wetlands in different countries.

The remainder of this paper is constructed as follows. Section 2 provides a background on urban wetland definition and types of wetlands. Section 3 explains the concept of biodiversity. Section 4 presents the effect of urban wetlands on urban heat islands. Section 5 defines the wetland and urban sustainability concept. Section 6 describes the recreational values provided by wetlands in the urban environment. Section 7 discusses the social perceptions about urban wetlands and how citizens contribute to conserving urban wetlands. Section 8 is a summary of the discussion and the research gaps in the urban wetlands studies. Finally, Section 9 concludes the paper.

2. Background on Urban Wetlands

According to the Ramsar Convention on Wetlands [6], urban and peri-urban wetlands are located inside and around urban areas and their suburbs. Wetlands can be either natural or artificial, i.e., constructed and can also be permanent or temporary, containing a low water depth of not more than 6 m of water. In principle, urban wetlands are classified as natural and constructed [25,26]. Natural wetlands (NWs) include rivers, lakes and their flood plains, swamps, estuaries, peatlands, tidal flats, coral reefs, and mangroves, while constructed wetlands (CWs) include artificially constructed canals, drains, reservoirs, artificial lakes, fish and shrimp farming ponds, ponds, rice fields, and stormwater treatment sites [27,28].

It is expected that, by 2050, the global population will increase to around 9.8 billion most of whom will live in cities [1]. Therefore, if urbanization is not planned and con-

trolled, the increased population growth in urban areas will be a threat to urban wetlands. The effects will include draining, contaminating, and destroying wetlands by land construction for housing, agriculture, and industry [29]. Unfortunately, due to the impact of urbanization, urban wetlands have become unconnected/fragmented. They have become patchy and distributed in different areas, and they have lost connectivity with each other. This habitat fragmentation in urban wetlands has led to a decrease in biodiversity in urban areas.

Urban wetlands possess a variety of ecological functions that cannot be replaced by other urban ecosystems. Urban wetlands are natural GBI in cities that host a wide range of biodiversity. Urban wetlands, due to their special role in urban ecological infrastructure, are known as a "city's kidneys" and "biodiversity library" [27]. In addition, the interaction with these valuable ecosystems in cities improves citizens' physical and mental health [30]. Urban wetlands offer a wide range of socio-cultural services, such as creating a space for recreation and leisure for the city inhabitants [31].

Urban wetlands not only provide ecological and recreational services in cities but also improve water quality by natural water purification and perform climate regulation. In terms of climatic regulation, urban wetlands create their own unique microclimate and reduce the overheating of urban environments [32]. As a result, the protection of urban wetlands is essential for obtaining sustainable living environments [12].

2.1. Urban Wetlands as Part of Green-Blue Infrastructure

Green-blue infrastructures (GBI) is presented as a strategy to deal with climate change in urban areas [33]. In urban developments, the GBI helps to optimize the use of lands in urban areas and meet the needs of people and nature in a sustainable way. In fact, GBI complements urban areas by combining hydrological networks with green areas and the built environments [34]. Among the diverse functions and advantages, in cities, the GBI enriches biodiversity, reduces global warming effects, enhances connectivity among ecological networks, and improves people's health and well-being [35,36].

In the urban context, the presence of GBI, such as trees, rivers, and ponds can increase thermal acceptability and establish climate-resilient urban systems [37], by obtaining higher cooling effects from blue infrastructures than the green infrastructures [38]. Moreover, if the area of GBI is larger than one hectare, there is a higher cooling effect (especially in summer) [39]. In cities, the GBIs have a significant therapeutic effect on human health and enhance the positive psychological reaction of humans [2,40]. Accessible GBI encourages physical activities and social cohesion and facilitates healthier living environments [41].

Within the GBI, the blue infrastructure can be either natural or constructed. Blue infrastructures, such as wetlands, contain waters that are easily accessible to humans in cities [42]. Wetlands are infrastructures that are composed of different elements, such as shore vegetation, soil, and water. This leads to a complex structure between land and water, harboring great biodiversity. Wetlands are transition zones that are caused by surface flooding or soil water saturation [27]. In principle, wetlands provide mechanisms to restore some of the natural processes that are needed to manage and create vital urban environments [43].

With proper planning and management, the ecosystem services of wetlands can benefit a growing urban population. Sustainable development and well-planned cities can generate higher levels of social welfare and drive economic growth and prosperity [44]. In sustainable cities, wetlands provide a range of ecosystem services, for example, by increasing water efficiencies, improving biodiversity, managing stormwater and flood regulation, regulating the regional climate, and mitigating climate change [45–47]. Wetlands also enable water quality protection, coastal protection, groundwater level and soil moisture regulation, and carbon sequestration [21].

Wetlands provide a large number of ecosystem services and the potential to be used as nature-based solutions to meet a variety of environmental, social, and economic challenges [48]. Unfortunately, during past decades, urbanization has changed the types or land use of wetland [49]. As a result, the reduction of wetlands has become one of the main threats to the sustainable development of urban areas [50]. It is mandatory to improve wetlands by sustainable developments and by applying and adopting appropriate methods for maintaining and protecting the existing wetlands or by constructing artificial wetlands in cities [46].

2.2. Urban Wetlands and Stormwater Management

Increased urbanization and subsequent human activities, such as uncontrolled construction are changing the watershed landscape. These changes increase runoff in urban areas, which has adverse effects on runoff quality and quantity, such as increased stormwater runoff pollutants [51]. Wetlands, whether natural or constructed, play an important role in cities for managing stormwater runoff including reducing the impacts of floods, absorbing pollutants, erosion control, groundwater recharge/discharge, and improving water quality.

In fact, urban wetlands can be an effective system for improving water quality. The complex hydrological, biological, physical, and chemical interactions that occur in a wetland lead to the natural reduction and effective purification of pollutants [52,53]. In addition, wetlands have an effective role in receiving storm runoff due to the inherent water storage and subsequently create a process to improve stormwater runoff quality. The most important wetland processes that improve stormwater runoff quality are sedimentation, filtration, adsorption and retention, ion exchange, precipitation, and biodegradation [51].

When water enters a wetland/pond through a stream or surface runoff, the water flows and passes through dense vegetation. The water flow velocity drops, and the suspended material in the water gradually settles to the surface of the wetland. Wetland plant roots can bind and remove up to 90% of the accumulated sediment from runoff or stream flow [52]. In addition, wetlands with dense vegetation, by reducing the velocity of the water flow through them, improve sedimentation and promote the removal of more contaminants.

The removal of contaminants by filtration through soils is effective in removing organic matter, phosphorus, bacteria, and suspended material. As runoff passes through the wetland, excess nutrients are absorbed by the wetland plants and accumulate in less harmful chemical forms [54]. When wetland plants die and decay, nutrients are recycled within the wetland. Wetlands are so effective in remove excess nutrients from stormwater runoff that it has led to the construction of wetlands specifically to treat effluent wastewater treatment in cities. It should be noted that natural wetlands are not suitable for this purpose, since there is a limitation for each wetland to how much can be added before the natural plant and chemical processes are overloaded and break down [55].

2.3. Natural and Constructed Wetlands

This subsection presents the definitions of natural wetlands (NWs) and constructed wetlands (CWs) and compares them. According to the United Nations Environment Program-Centre on Water and Environment (UNEP-DHI) (https://www.unepdhi.org/, accessed on 15 September 2021), "Natural wetlands are ecosystems that are permanently or seasonally saturated in water and create habitats for aquatic plants and provide conditions that lead to the development of hydric soils" [28,56,57].

NWs are GBI systems that can be found in diverse geo-environmental settings around the world [46]. NWs consist of rivers, lakes, saltwater lake, estuaries, swamps, tidal flats, coral reefs, peatlands, and mangroves [27]. In all environmental settings, NWs refine and enhance the quality of water passing through the system, because they operate as ecosystem filters [35]. In the other word, NWs are understood as high-efficiency disinfecting ecosystems. In particular, in local environments, NWs provide ecological flood protection and clean water [58]. They also offer shelters for birds during breeding and feeding [59], and preserve more native plant species [55].

In contrast to NWs, the constructed wetlands (CWs) are defined as "man-made complexes of the saturated substrate, emergent and submerged vegetation, animal life and water that simulate natural wetlands for human use and benefits" [60]. CWs contain stormwater basins, constructed canals, drains, reservoirs, artificial lakes, fish, and shrimp farming ponds, constructed ponds, rice fields, and sewage treatment sites [27,57]. CWs are ecologically engineered systems that have similar functions to NWs. CWs are used as an alternative cost-effective approach to conventional wastewater treatment [61,62].

CWs are affordable and sturdy systems that are low cost, easy to maintain, and easy to operate systems that only need periodic on-site labor [35]. In general, CWs are the repetition of the natural process of NWs that pursue a beneficial purpose. This means that CWs are constructed to emulate and improve the function of NWs [63]. Sometimes CWs can be constructed for specific purposes (such as flood control and surface water management) to support a specific environmental concern and provide sustainable environments [64,65].

In living environments, CWs are accepted as a practical and effectual approaches to improve the environmental quality of cities by having a key role in revitalizing urban ecology [66]. For example, CWs utilize natural processes that are suitable to remove pollutants from contaminated water within a more controlled environment [67]. Moreover, CWs provide habitat and biodiversity, support recreational activities (such as bird watching), storing water during periods of drought and saturation, and adding aesthetic value in urban areas [68].

Both NWs and CWs can be considered as an alternative to conventional systems for wastewater treatment. Both systems contain vegetation, substrates, soils, microorganisms, and water. They use multiple processes, including physical, chemical, and biological mechanisms, to eliminate different contaminants and subsequently enhance the quality of the outlet water. Indeed, comparative studies considering the ecological operations of natural and constructed wetlands indicate that both accomplish relatively similar ecological functions [35,69].

Despite containing similar constructing elements, NWs and CWs have significant differences in their intended use and functions. For example, while CWs are used for the purifying of contaminated wastewater in urban areas [70]. NWs are normally not used for wastewater treatment purposes as this can yield irreparable detriment to the ecosystem of these wetlands [71].

A study by Rooney et al. [55] introduced three structural and biophysical differences between NWs and CWs. First, while CWs are usually steep-sided, the shores of NWs have much gentler slopes. Second, NWs are usually more strongly connected to both surface and groundwater streams compared with CWs. This is due to the fact that CWs are often covered with clay to prevent any connection to the groundwater, and their water levels are often maintained at constant depths. The third difference is related to the difference in landscape positions of NWs and CWs. While NWs are naturally created in different environments, CWs are constructed in peri-urban areas with higher population density, locations with higher exposure to roads and contaminants, and locations with impervious beds.

Another difference between CWs and NWs refers to the hosting biodiversity, as NWs have considerably more habitat types than CWs [72]. Indeed, while NWs are commonly a habitat for native species, CWs often host non-native species and have the potential to increase the number of undesirable species [55]. Still, due to the diverse benefits offered by natural and constructed wetlands, both need to be preserved and have their functionality improved. This could be done by preserving natural wetlands and stopping their loss as well as constructing new wetlands in urban areas.

Along with urbanization, the number of CWs have increased by 233% from the year 1970 to 2014 [73]. However, due to diverse human activities, including the expansion of urbanization, agriculture, and aquaculture which has taken place over the last decades, the majority of NWs have been significantly manipulated, destroyed, fragmented, or totally lost [74]. A research report that more than half of the world's NWs have disappeared

during the last century [75]. Another study supported this finding by considering more than 2000 wetlands around the world and reported that the number of NWs decreased by an average of 35% from the year 1970 to 2015 [73].

3. Urban Wetlands and Biodiversity

Wetlands are biologically diverse systems that improve water quality and sequester carbon [76]. As significant biodiversity sources, wetlands provide habitats for groups of species from micro-organisms to mammals [77]. Examples of these species include amphibians, insects, reptiles, birds, and mammals (e.g., beavers) that are uniquely adapted to aquatic environments [78,79]. Indeed, wetlands increase the biodiversity in urban areas by acting as networks of fragmented habitat to facilitate the movement of species in the environments [80,81].

Unfortunately, due to the urbanization and the development of urban areas, wetlands as habitats have been fragmented. Fragmentation of wetlands indeed damages the habitat and has become a major challenge in urban environments [82]. Although the fragmentation of wetlands is a major threat to their existence, they remain important and are highly functional for wildlife species [83]. Therefore, identifying the importance of wetlands, preserving them, and possibly increasing the connectivity between them would considerably support the protection of biodiversity in urban areas [84].

Even preserving wetlands that are considered of lower quality (in terms of reduced biodiversity) and polluted (in terms of water quality) has numerous advantages compared to the situation of totally lacking wetlands or having fewer of them [85]. This is because, when fragmentation of urban wetlands occurs, low-quality habitats can play an important role by supporting connectivity between good patches. In this way, a sub-optimal habitat network structure can support a higher level of biodiversity on a landscape level [82].

Due to the significance of wetlands for providing habitats and supporting biodiversity in urban areas, in the following, we provide a review on this topic and summarize the objectives, methodologies, and findings of the reviewed articles in Table 2.

Reference	Objective	Methodology	Findings
Melbourne, Australia (Hale et al., 2019) [82]	Highlighting the potential ecological effects of stormwater wetlands to manage the unintended consequences for urban biodiversity	Investigated 67 urban wetlands with pollutant concentrations to specify storm wetlands could be ecological traps for native amphibian and fish in the studied areas	The stormwater wetlands often become habitats for animals, which is beneficial for the persistence of species in cities
Vihti, Finland (Wahlroos et al., 2015) [69]	Designing two wetlands with slightly different and monitored them for 5 years	Studied the vegetation establishment, water quality improvement, animal settlement, as well as people's recreation	In the second year, vegetation was self-established and wetlands became successful breeding grounds for amphibians and birds and offered recreation values to people
Netherlands and New Zealand (van Roon, 2012) [86]	Investigating the role of wetlands in carbon sequestration and evaluating biodiversity loss in the urbanization process	Used the literature review and case study investigation in a period from 2002 to 2010	There are problems in creating suitable conditions for a variety of rare and vulnerable wetlands near urban use
Melendugno, Italy (Semeraro et al., 2015) [87]	Assessing the role of multifunctional CTW in terms of biodiversity and enhance ecosystem services	Monitored fauna and flora, preparing habitat map by GIS	CTW's ability to provide side benefits beyond the main purpose of water treatment, conservation of wildlife habitats and biodiversity

Table 2. Urban wetlands and biodiversity.

Reference	Objective	Methodology	Findings
Helsinki, Finland (Liao et al., 2020) [88]	Examining how urbanization influences the diversity of diving beetles	Sampled diving beetles in 25 urban ponds using the GLMM model	The model revealed that urbanization reduced the richness of diving beetle species but had little effect on their abundance
Catalonia, Spain (Gascon et al., 2009) [89]	Conducting conservation biology by prioritizing sites based on high biodiversity	Regression tree models were used to identify key factors affecting biodiversity, including water, wetland, and landscape features as explanatory variables	The biodiversity criteria used in this study were significantly related to some explanatory variables. Significant positive relationships were found between some biodiversity criteria and wetland habitat conditions
Guapore, Brazil (da Silva et al., 2015) [90]	Investigating development targets and planning tasks for the area between the Pantanal and the Amazon as an important ecotone or transition zone	Used the (DPSIR) framework to evaluate cause-and-effect relationships	Planning and management in this wetland in three ways: (1) Business as usual (2) Conservation actions (3) Integrating biodiversity objectives into other policies and planning strategies
Meli et al., (2014) [91]	Presenting a meta-analysis to evaluate the effectiveness of ecological restoration and identify what factors influence	A literature review was conducted to identify quantitative studies on the effects of ecological restoration	The meta-analysis study showed that ecological restoration increases biodiversity and ES supply
Lombardy, Italy (Morganti et al., 2019) [92]	Studying the bird communities of inland wetland	Environmental variables were collected at the two different spatial scales of Natura 2000 sites and point counts respectively	The extent of the reedbeds/mires was positively associated with the occurrence of all species of conservation concern at the site scale
Andalusia, Spain (Guareschi et al. 2015) [93]	Exploring the relationships between community composition and species richness of waterbirds and aquatic macroinvertebrates in 36 Ramsar wetlands	Waterbird data surveys, as part of an official monitoring program, were performed by the Regional Government	The collection of waterfowl was more affected by climatic variables and water levels, while conductivity was the most important factor affecting large vertebrate communities

Table 2. Cont.

A study by Hale et al. [82] highlighted the potential ecological effects of stormwater wetlands to manage the unintended consequences for urban biodiversity. The study investigated 67 urban wetlands with pollutant concentrations to specify whether storm wetlands could be ecological traps for native amphibians and fishes in the studied areas. The findings of this study stated that the stormwater wetlands often become habitats for animals, which is beneficial for the persistence of species in cities. Another important finding is that the animals that colonise the stormwater wetlands suffer from the accumulated pollutants.

Based on these findings, this study highlighted the following key considerations for stormwater wetland management to reduce its negative effects on biodiversity. The accumulation of pollutants and adverse effects on amphibians and other animals is one of the main aspects of habitat quality in relation to storm wetlands. Therefore, it is suggested that inspection and maintenance programs be considered to ensure the function of storm wetlands. Another consideration pertains to the ecological consequences of changes in wetland quality. Changes in the quality of wetlands can cause ecological traps, which are recognized as an unintended consequence of management activities. Ecological traps are usually a serious situation, but they remain hidden and unknown.

Wahlroos et al. [69] evaluated the design of two urban wetlands with slightly different designs in urban parks. The two wetlands were designed to adapt open water areas for habitat and recreation at the cost of densely vegetated areas. The two wetland parks were designed to have sufficient wetland space for amphibian habitats. Larger open water areas, as well as islands, were designed as habitats for both wetland parks to provide waterfowl habitats and attract people. The study showed that, in the second year, the vegetation was self-established.

The vegetation establishment reached 102 species with 97% native plants after 5 years. Furthermore, the results of wildlife observation showed that breeding of amphibians and water birds was successful after constructing the wetlands. These wetlands also became successful breeding grounds for spawning amphibians and nesting birds. Thus, the wetlands succeeded in creating high biodiversity at the habitat scale in the center of a residential community. Moreover, the study reported the recreation values of peoples' everyday visits due to the increase of biodiversity and vegetation in these parks in the city of Vihti.

Van Roon [86] investigated the role of wetlands, such as bogs, fens, and swamps in carbon sequestration and evaluated the biodiversity loss in the urbanization process. This study reviewed the literature related to historical degradation, current maintenance, and management of wetlands, including bogs, fens, and swamps. Additionally, Van Roon investigated these sites in the period from 2002 to 2010, analyzed the documents related to the site, and interviewed staff from the site information centers as well as municipal planners.

Based on the literature review, this study concluded that creating suitable conditions for the reconstruction and maintenance of vulnerable wetlands is very difficult for swamps to fens to bogs near urban areas. Creating these conditions requires minimizing air emissions and manipulating groundwater flows, protecting springs, and minimizing nutrient depletion through the surface or groundwater. For instance, bogs survive in the lowest-density urban development areas.

Ecological corridors that contain fen wetland remnants can survive in development areas only with high biodiversity. In fact, fens survive throughout the ecological corridors near high-density urban areas, but the results showed that they are chemically and hydrologically degraded, and their contribution to stopping biodiversity loss is limited. Furthermore, achieving these conditions helps water-centric development and corridor reservations and is beneficial to all stakeholders.

Semeraro et al. [87] aimed to assess the role of constructed treatment wetlands (CTW) in terms of biodiversity and enhanced ecosystem services. This study used annual monitoring of fauna and flora to identify national and international species strongly related to available new habitats. In the first stage, to identify the CTW wetland habitat, a habitat map was prepared by taking photos and orthophotos and then classifying the habitat using the Commission of the European Communities, 1991 (CORINE) habitat classification.

The habitat map was validated and updated through inspections and field surveys at GIS. The second stage was done by describing the vegetation to identify different types of plant communities in the basins and canals, along the beaches, in artificial soils, and in the garrigue. The outcomes of the study confirmed CTW's ability to provide side benefits beyond the main purpose of water treatment, such as the conservation of wildlife habitats and biodiversity at local and international scales, as well as its ability to create recreational and educational value.

Liao et al. 2020., [88] examined how urbanization influences the diversity of diving beetles (Dytiscidae) and the effect of pond margin steepness, as well as the presence/absence of fish in the pond on urban diving beetles. In this study, diving beetles were sampled using activity traps in 25 urban ponds (14 ponds without fish and 11 ponds with fish). In the study, various characteristics were considered, such as the pond water depth, pond size, shoreline length, and proportion of impermeable surface in a buffer zone.

The results reveal that urbanization reduced the richness of diving beetle species but had little effect on their abundance. This indicates that urbanization does not diminish the capacity of ponds to support diving beetle species, as their numbers are unchanged; however, some species react negatively to urbanization. The presence of fish in the ponds compared to the absence of fish has a very significant and negative effect on species richness.

The presence of fish had a stronger effect on the richness of diving beetle species compared with urbanization and the pond margin steepness. Furthermore, the pond margin steepness had no statistically meaningful effect on the richness of diving beetles in ponds without fish. However, the interaction between the pond margin steepness and the presence of fish had a very notable and negative effect on diving beetles.

A study by Gascon et al. [89] aimed to identify the key factors affecting the biodiversity in wetlands to find a relationship between biodiversity metrics, conservation status, and habitat conditions. The objectives of the study were:

- (*i*) comparing the reactions of different biodiversity metrics,
- (*ii*) recognizing key environmental factors for different biodiversities, and
- (*iii*) investigating whether wetlands with high biodiversity also have good habitat conditions and high protection status.

In this study, 91 wetlands (such as ponds, lagoons, and marshes) were sampled at the assemblage level (crustaceans and insects). The study used regression tree models to identify key factors affecting biodiversity. Thus, the study used variable factors, including the dissolved inorganic nitrogen, soluble reactive phosphorus, total nitrogen, total phosphorus, chlorophyll-a, conductivity, water permanence (temporary vs. permanent), water body size, wetland isolation, and water body density. The study calculated eleven biodiversity metrics, such as the assemblage structure, rarity, and taxonomic distinctness for each (crustacean and insect) sample. Among the eleven metrics, three metrics were related to the structure of the assemblage, including:

- (*i*) the number of species in each sample,
- (ii) the species diversity obtained using the Shannon–Wiener diversity, and
- (*iii*) Pielou's evenness (species evenness) based on Shannon's index.

Analyzing the key factors determining the biodiversity of wetland aquatic invertebrates, the results showed that five of the eleven biodiversity metrics used in this study were significantly related to some explanatory variables. Moreover, the results obtained from the comparison of the two sampled seasons (winter vs. spring) showed that conductivity was the main factor influencing biodiversity metrics. Significant positive relationships were found between certain biodiversity metrics and wetland habitat conditions, while there was no case for conservation status, indicating the inadequacy of conservation policies to protect aquatic invertebrate biodiversity.

A study by da Silva et al,. [90] investigated the development targets and planning tasks for the Guaporé–Paraguay wetland, which is an area between the Pantanal and the Amazon as an ecotone with high biodiversity importance. It is worth noting that an ecotone indicates a transitional area of vegetation between two different plant communities, such as forests and wetlands. The study used a framework named the driver pressure state impact response (DPSIR) to evaluate cause and effect relationships between the interrelated components of social, economic, and environmental systems.

These interrelated components include the driving forces of environmental change; pressures on the environment; state of the environment; impacts on population, economy, ecosystems; and the response of the society, e.g., policy response. Note that the DPSIR approach was originally derived from the social sciences and later became extensively accepted as a general framework for organizing information about the state of the environment.

This research utilized a database of plant and animal species including the presence/absence information, abundance, and diversity index for different scales. Then, they analyzed the existence and distribution of plants, mammals, birds, fish species, macrophytes, peri-phytons, and zooplankton in order to assess the biodiversity status of the region. As a result, the research proposed the following three strategies for planning and management of the Guaporé–Paraguay ecotone:

- (i) Business as usual, which refers to a further decrease of natural areas. The court of justice decided that Guaporé–Paraguay does not require special protection in the state planning system. Thus, this strategy will result in ongoing forest and river fragmentation.
- (ii) Conservation actions that calls for the restoration of riparian deforested or degraded areas and protecting wetlands in both basins. The development of conservation actions can lead to the expansion of current protected areas and management plans in the region; therefore, regional protected areas can be identified to preserve a large area of river forests to survive the priority species of the Guaporé–Paraguay ecotone.
- (iii) Integrating biodiversity objectives into other policies and planning strategies, which refers to the restoration of riparian deforested or degraded areas and the protection of wetlands in the basin. This strategy integrates biodiversity goals in the planning and implementation of hydroelectric dams and agricultural management.

A study by Meli et al. [91] reviewed 70 experimental studies to identify quantitative studies on the effects of ecological restoration on the biodiversity and ecosystem services of degraded aquatic and semi-aquatic wetlands. A meta-analysis identified the factors influencing restoration. The study compared the performance factors of the selected ecosystems between (1) the destroyed and restored wetlands; and (2) between the restored and natural wetlands using response ratios and stratified modeling of random effects.

The meta-analysis showed that ecological restoration increases biodiversity and ecosystem services supply in degraded wetlands and, thus, benefits the human communities that interact with them. The exact effects of wetland restoration strongly depend on the underlying factors, thus, emphasizing the need for specific habitat planning and evaluation of restorations. Furthermore, biodiversity demonstrates good recovery, although the exact recovery strongly depends on the species.

Restoration wetlands showed 36% of ES supply, regulation, and support levels compared to degraded wetlands. The biodiversity recovery and ecosystem services also positively showed a correlation, which represents an effective restoration result. Moreover, the restored wetlands showed a level of ecosystem services similar to natural wetlands.

Morganti et al. [92] studied the bird communities of an inland wetland. This study aimed to:

- (*i*) understand the landscape-scale variables affecting the biotope level occurrence of conservation birds,
- *(ii)* identify the habitat variables related to the occurrence of a set of target reedbeddwelling species, and
- *(iii)* achieve practical management recommendations for the protection of bird communities and populations in the inland wetlands.

The results showed that the extent of the reed beds/mires was positively associated with the occurrence of all species of conservation concern at the site scale. At the field scale, the reed bed extent positively predicted the species' occurrence but only in the presence of patches of clear shallow water. Species-specific MARS models qualitatively demonstrated similar results for some species but were generally outperformed by multi-species.

Guareschi et al. [93] explored the relationships between the community composition and species richness of waterbirds and aquatic macroinvertebrates in 36 wetlands. As core objectives, this research aimed to:

(i) test the congruence of the patterns of species composition and richness among waterbirds and aquatic macroinvertebrates, and (*ii*) investigate which environmental variables were associated with the biodiversity patterns of waterbirds and macroinvertebrates in order to identify the key factors explaining potential discordance in these patterns.

The study demonstrated that climatic variables and water levels mostly affected the collection of waterbirds; while conductivity was the most important factor affecting large vertebrate communities. The results depict a slightly inverse relationship in the richness patterns, where wetlands that are rich in waterbird species are less rich in Hemipetra families and macroinvertebrates. The results of the linear models also demonstrate that, in general, different environmental variables were related to the richness patterns of different classification groups. In addition, the analysis of different biological communities revealed that using datasets of different classification groups is an essential prerequisite for successful policies and monitoring of wetland conservation. The research concluded that there is a need for creating a diverse and complete network of protected sites, which can maintain multiple biodiversity components in wetlands.

To conclude the section, wetland biodiversity has been severely disrupted as a result of urbanization, as urban development is a primary factor in reducing the biodiversity of wetlands. In the literature, the studies explain that, when natural or human factors destroy wetlands, ecological restoration is often performed to preserve biodiversity and ecosystem services. Consequently, the preserved wetlands become a breeding ground for wildlife and strengthen the biodiversity in wetlands.

Wetlands create a network of fragmented habitats and provide feeding, spawning and nursing areas for many species, such as invertebrates, amphibians, birds, and fish. Preserving biodiversity in wetlands is essential to maintaining the vital functions of wetland ecosystems and preserving the values they provide to their environment. The maintenance of biodiversity in wetlands also can be achieved by raising public awareness, which requires continuous guidance and learning at the public level.

4. Urban Wetlands and Urban Heat Islands

In recent decades, urbanization associated with the increase in human activities has changed the land use, land cover, and hydrological cycles. The increase in human activities has impacted the microclimate of urban areas through the increase in urban temperature, known as the Urban Heat Island (UHI) effect [94–96]. This effect is one of the well-known effects directly related to the current climate change [97,98]. The UHI effect appears when an urban area is significantly warmer than the surrounding areas [94,99]. The reason for the UHI effect is the change in the type of the land surface use [100]. When human-made surfaces replace natural landscapes, the latter surface can store additional heat that is slowly released at night [101–103].

UHIs influence the energy consumption, environmental quality, and human health in urban areas [101]. Hence, the relationship between land use and urban climate has increasingly attracted the attention of urban planners and policymakers to coordinate development strategies to boost the well-being of people [4]. In this vein, the natural and eco-friendly cooling effect of GBI has been realized as a solution among research communities, strategists, and urban planners [104].

However, regardless of the cooling effect of water bodies, e.g., urban wetlands, in practice, they have been less investigated compared to green spaces, such as forests and parks [39]. This is one of the motivations behind our study. Indeed, urban wetlands are areas located inside or at the borders of cities where they offer a variety of benefits and ecosystem services, such as their potential for providing a significant cooling effect (CE) to surrounding areas and mitigating the UHI effects on urban areas [101].

To study the impacts of UHIs on the microclimate of urban areas and understand the methodologies and approaches for mitigating UHI impacts through the cooling effects of urban wetlands, we conducted a review on the topic. We summarize the objectives, methods, and findings of the reviewed articles in Table 3.

Chun-ye and Wei-ping [65] aimed to analyze the effect of urban wetlands on city temperature by using remote sensing technology. This study used Landsat TM remote sensing images from the city of Hangzhou and applied an algorithm to analyze the distribution of the heat island effect qualitatively. They demonstrated that urban wetlands well mitigate the urban temperature. This temperature regulation was more significant near urban wetlands.

In addition, the heat island intensity gradually decreased from the periphery to the urban area of Xixi Wetland. The higher distance from the wetland indicates a greater average surface temperature and weaker wetland regulation. Furthermore, the range of temperature regulations and the extent of urban wetlands on UHIs can change due to the types of wetlands. This change can include the ratio and pattern of water distribution in urban wetlands, vegetation and other types of landscapes.

Reference	Objective	Methodology	Findings
Beijing, China (Sun et al., 2012) [105]	Evaluating the cooling effects of wetland in urban areas	15 urban wetlands were assessed using the ASTER images technique	The Landscape Shape Index and the location of wetlands are remarkable indicators that impact the intensity of UCI in an urban area
Changchun and Jilin, China (Xue et al., 2019) [101]	Quantitatively evaluating the cooling effect of wetlands	Used remote sensing data in the form of Landsat-8 TIRS with two thermal infrared spectral bands to regain LST	The CE of urban wetlands is considerably associated with the wetland area, shape, and hydrological connection
London, UK (O'Malley et al., 2014) [106]	Investigating minimizing UHI effects in London, UK.	Simulations used ENVI-met software, using the Urban Futures Assessment Method	Constructions and buildings forms orientation and layout are highly effective UHI mitigation strategies
Shenzhen and Hong Kong, China (Lin et al., 2020) [107]	Measuring the amount of surface urban heat island (SUHI) reduction, that induced by water bodies in the urban environment	Used the Google Earth engine to map the severity of SUHI and water bodies	A 10% increase in water body surface was accompanied by a 11.33% reduction in SUHI severity
Sivas, Turkey (Kuscu simsek and Odul, 2018) [13]	Determining the distance at which wetlands exhibit a cooling effect as well as revealing the impact of land use change	Used remote sensing techniques and collected four consecutive satellite images	The regular temperature increases up to a maximum of 1000 m around the wetland
City model (Gunawardena et al., 2017) [4]	Reviewing the effectiveness of green and blue spaces in decreasing the risks of heat-related illness from high temperatures of the cities	Used two city models, including "compaction" and "dispersal"	The compaction model increased the strength of UHI, the dispersal model decreased the UHI impact
City model (Theeuwes et al., 2013) [108]	Investigating the effect of blue spaces on the urban temperature and human thermal comfort	Used the WRF model to consider the surface water coverage, size, spatial configuration, and temperature variations	A large lake has a great impact on its surrounding temperature and thermal comfort, while several smaller lakes affect a higher percentage of the city and offer better human thermal comfort

Table 3. Urban wetlands and urban heat islands.

Reference	Objective	Methodology	Findings
Sheffield, UK (Hathway and Sharples, 2012) [109]	Investigating the effectiveness of small urban rivers in reducing the impact of UHI.	The temperature and humidity were measured in 12 places near the river by Ibuttons (Maxim, San Jose, CA, USA) measurement devices using 12 plate Gill solar radiations screens	Significant cooling over the river with an average cooling of nearly 1 °C when the ambient temperature was higher than 20 °C
Hangzhou, China (Chun-ye and Wei-ping., 2011) [65]	Analyzing the effect of urban wetland on city temperature	Used Landsat TM remote sensing images	Urban wetlands have good regulation on urban temperatures
Rotterdam, Netherlands (Steeneveld et al., 2014) [110]	Investigating the relation between UHI and open water fraction (OWF)	Used googlemaps [™] to map the relation between the UHI and OWF	Blue spaces have a relatively high heat capacity compared to their surroundings

Table 3. Cont.

Sun et al. [105] evaluated the cooling effects of wetlands in urban areas. This research used ten reservoirs and lakes and five rivers as wetlands to assess the severity of the UCI. This research aimed to:

- (*i*) develop a method for estimating the UCI severity of wetlands,
- (*ii*) evaluate the effects of the area, forms, and sites of wetlands on the UCI severity, and(*iii*) discuss the importance of evaluating ecosystem services and landscape design in urban areas.

The selected wetlands were evaluated using the ASTER (advanced spaceborne thermal emission and reflection radiometer) imaging technique to obtain cooling effects. This technique was used because air temperature is highly associated with the land surface temperature. The findings of this research stated that the Landscape Shape Index (LSI) and the location of wetlands are remarkable indicators that impact the intensity of UCI in an urban area. It is also essential to evaluate the quantification of a wetland microclimate regulation and urban landscape design to reduce the effects of UHI.

A study by Xue et al. [101] evaluated the cooling effect of 21 urban wetlands and three green spaces in cities. This study used remote sensing data in Landsat-8 TIRS with two thermal infrared spectral bands to obtain the LST and measure the selected areas' UHI reduction abilities. Then, the study performed correlation analysis to examine the relationships between the characteristics of wetlands and surrounding buildings and two cooling-effect indices: the normalized cooling capability index (NCCI) and normalized cooling efficiency index (NCEI).

The results showed that the cooling effect of urban wetlands was considerably associated with the wetland area, shape, and hydrological connection. The cooling effect was better in wetlands with more complex shapes. The average NCCI of wetlands connected to other surface waters was six-times higher than separated wetlands. Notably, there was an inverse relationship between the cooling capacity of urban wetlands and the height and density of surrounding buildings. The size of wetlands was the most important element to reduce the air temperature. More extensive wetlands had a more significant cooling effect compared with smaller wetlands. As a result, to increase the cooling effect of urban wetlands, wetlands should be designed more naturally while considering the surrounding built areas.

O'Malley et al. [106] investigated minimizing urban heat island (UHI) effects. UHI is a significant problem in the UK that requires reduction. They evaluated effective and resilient UHI mitigation strategies to guide architects, urban designers, urban planners, and city authorities. First, the research defined four possible future scenarios for 2050 to conducted the UHI mitigation strategy. Then, they performed simulations using ENVI-met

software and input the parameters from the surrounding area, such as tall trees with dense crowns and dense hedges.

The research also utilized the Urban Futures Assessment Method (UFAM) to assess the resilience of the mitigation strategies. The study stated that the construction and building forms, orientation, and layouts were highly effective UHI mitigation strategies. These layouts may include the use of trees, shrubs, grass, and high albedo materials in external building surfaces. The assessment also showed that the use of these layouts had a similar level of resilience.

A study by Lin et al. [107] measured the amount of surface urban heat island (SUHI) reduction induced by water bodies in the urban environment. The study area was in the middle of the Pearl River Delta (PRD) metropolitan area. This research aimed to:

- *(i)* quantify the cooling effect of water bodies,
- (*ii*) compare water bodies regarding the thermal context, and
- (*iii*) identify the impact of water body overflow in reducing the SUHI phenomenon.

In this study, Google Earth was used to map the severity of SUHI and water bodies to investigate water bodies' cooling effect and efficiency in a metropolitan area. The study determined that water bodies with different features had a considerable cooling impact on the local thermal environment. In addition, a 10% increase in water body surface was accompanied by a 11.33% reduction in SUHI severity.

A study by Kuşçu Şimşek and Ödül [13] investigated the distance at which wetlands exhibit a cooling effect and revealed the impact of land-use change. As a methodology, they selected five wetlands. Then, they used remote sensing techniques and collected four consecutive satellite images in the summer. The temperature values were taken from Landsat TM-5 images on 18 June, 4 July, 20 July, and 5 August in 2007. Then, they analyzed their images utilizing GIS software. In the analysis phase, they investigated the correlation between surface temperature information and land use information as well as the distance from shoreline information.

This research used the following three phenomena: permanently irrigated land, which means forest areas that include wetlands; non-irrigated arable land; and pastures, which refers to open spaces with little or no vegetation with water bodies. The research used one "permanently irrigated land", one "non-irrigated arable land", and one "pasture"; and studied the cooling effects of their wetlands at 1000, 900, and 900 m distances, respectively.

As a result, they found that the regular temperature increased up to a maximum of 1000 m around the wetland. The maximum temperature difference concerning the distance was 16.29 °C in 400 m of "non-irrigated arable land". Another important observation was the temperature change in different areas with different land uses. The type of land cover is expected to have a significant impact on the different land uses. The study also emphasized the necessity of climate-sensitive land use planning. This research suggested that the land use around wetlands must be decided with awareness. This is to enhance the climate regulatory effects of wetlands. The efficient use of natural resources is important to identify the strategies to adapt to climate change.

Gunawardena et al. [95] conducted a literature review on the effectiveness of green and blue spaces in decreasing the risks of heat-related illness from high temperatures in cities. In this research, green space referred to areas covered with vegetation, and blue space indicated water bodies or watercourses in the cities. This research provides critical notes useful for urban planning and urban development. The study used two city models, including "compaction" and "dispersal". To this end, the research used these models to discuss the impact of UHI on cities. This literature review showed that the compaction and dispersal urban models change the mesoscale UHI in different ways.

While the compaction model increases the strength of UHI, the dispersal model decreases the UHI impact. Nevertheless, due to the lack of enough evidence, it is difficult to conclude that one urban model is superior to another in terms of heat balance optimization. Another result explains that different parameters, such as scale, geometry, extent, and distance of interventions, surface roughness, fetch length, the morphology and material of

the context, and the prevailing climate impact the thermal effects in blue and green spaces in urban areas.

Greenspace extends the micro-scale cooling effect, while blue spaces may provide a warming effect when the UHI intensity and the risk of heat stress are higher (especially at night in late summer). When green and blue operate together, they create a mutually dependent environment and offer synergistic cooling.

A study by Steeneveld et al. [110] investigated the relation between UHIs and the open water fraction (OWF). The study used the data collected from weather observations and a weather monitoring station. Then, they investigated the relation between UHI and open water fraction (OWF) utilizing the collected dataset. To map the connection between the UHI and OWF, they used Google Maps[™] images for a range of possible buffer sizes. The results showed that the UHI was significantly related to the OWF.

Physically, blue spaces have a relatively high heat capacity compared to their surroundings. In late summer and autumn, the blue spaces gained a relatively high temperature compared to rural areas. The results also showed a considerable increase of the 95 percentiles of the UHI with open water surface availability.

Theeuwes et al. [108] conducted a quantitative study regarding the effect of blue spaces on urban temperature and human thermal comfort. This research studied how the spatial distribution of blue spaces and water fractions in the city impacts the urban temperature. It also quantified the impact of the lake water temperature in urban areas and thermal comfort. The research used a weather research and forecasting (WRF) model to evaluate the influence of surface water on urban temperature. This model utilized the impact of geometry of the street canyons and shadowing from built environments.

The model computes the energy exchange between the atmosphere and urban facades, such as roofs, walls, and roads. Using the WRF model, the research takes into account the surface water coverage, size, spatial configuration, and temperature variations. The results showed that the cooling effect of water depends nonlinearly on the fractional water cover, size, and distribution of individual lakes within the city. The air temperature change due to lakes depends on the size, the number of lakes, the distance from the lake, and the water temperature. In general, the cooling effect of water decreases with distance from the lake.

However, the impact of the lake is still measurable several kilometres downwind of the lake. A large lake has a significant effect on its surroundings temperature and thermal comfort, while several smaller lakes affect a higher percentage of the city and offer a better human thermal comfort. As a result, proper weighing of the sizes and locations of the blue spaces is helpful for urban planning.

Hathway and Sharples [109] investigated the effectiveness of small urban rivers in reducing the impact of UHI. They also explored the role of urban riverbanks in reducing the cooling effect. Hence, they perform a study along the River Don. They measured the temperature and humidity in 12 places near the river or vertical to the riverbank. Therefore, they mounted Ibuttons (Maxim, San Jose, CA, USA) measurement devices in 12 plate Gill solar radiations screens in those places.

The measurement results showed significant cooling over the river with an average cooling of nearly one °C when the ambient temperature was higher than 20 °C. The results also demonstrated substantial cooling by almost two °C and 1.5 °C cooling on the riverbank on hot days in May. The banks, which included engineering materials had a higher temperature than the banks having vegetation. In addition, open street intersections with rivers had more cooling effects than streets shut off from the river.

To summarize this section, urban wetlands play an important role in reducing the effects of urban heat islands through their cooling effects. The reviewed articles explain that the effect of urban wetlands regulating the urban temperature is very high. While, closer to wetlands, the higher the regulation effect. Indeed, the cooling effect of urban wetlands brings positive benefits, especially for improving environmental health. However, the cooling effect also is significantly affected by the shape and size of the wetland, the

hydrological connections of wetlands, the surrounding built areas, and the combination of wetlands with vegetation.

Therefore, it is necessary to consider these factors in urban wetland construction, maintenance, and rehabilitation. The proper and optimal wetland design to increase the cooling effects is of vital value in urban development. Considering the cooling effect of wetlands in urban design and development is critical when guaranteeing urban sustainability in the era of climate change.

5. Urban Wetlands and Urban Sustainability

With the increase in urbanization, proper planning and management of natural resources can contribute to urban sustainability. Sustainable cities contribute to economic growth and prosperity and are examples of sustainable development [44]. In sustainable urban areas, wetlands as natural resources provide various ecosystem services, such as increasing the biodiversity, improving water quality, and mitigating climate change [45]. Furthermore, wetlands can preserve human safety by stormwater management in these cities and provide a comfortable living environment by regulating regional climate [46].

However, in past decades, urbanization has threatened wetlands by destroying and changing the land use from GBI to residential or industrial buildings. As a result, the decline in natural wetlands has become one of the central warnings to preserving these valuable natural resources and contribute to urban sustainability [50]. Thus, appropriate approaches should be adopted to evaluate and analyze the sustainable development of wetlands [46].

Wetlands are known for providing ecosystem services and, therefore, have excellent potential to be used as nature-based solutions to meet various environmental, social, and economic challenges [21]. To study the impacts of wetlands on urban sustainability and learn the practices of preserving wetlands in urban areas, in the following, we present a literature review on this topic and summarize the objectives, methodologies, and findings of the reviewed studies in Table 4.

Roy-Basu [111] reviewed the East Kolkata Wetlands (EKW) and suggested a sustainable model. The EKW is one of the largest aquaculture systems where municipal wastewater is recycled for aquaculture and agriculture. The wetlands are also used for flood mitigation in the city of Kolkata. Indeed, in recent decades, rapid encroachment on the wetlands has had various environmental, social, and economic impacts due to unplanned development.

Reference	Objective	Methodology	Findings
Panjin, China (Su et al., 2020) [112]	Identifying the ecosystem value and sustainability of wetland	Used Emergy analysis	Human activities, developments, and investments in the area play critical roles in protecting wetland ecosystems
Jecheon, Korea (Byeon and Nam. 2020) [113]	Investigating the treatment of non-point source (NPS) pollution and biodiversity increase in a constructed wetland	Used an SSB system as a treatment wetland for NPS pollution with short retention time and high biodiversity	SSB wetlands indicated a treatment impact on NPS pollution at the initial rainfall events within the moderately short retention time
Shanghai, China (Wu et al., 2018) [114]	Addressing the effects of wetland rehabilitation on the coastal environment	Used an object-oriented classification method, and digital shoreline analysis system	Considering the resilience of wetland ecosystem, a balance should be established between land use and wetland rehabilitation

Table 4. Urban wetlands and urban sustainability.

Reference	Objective	Methodology	Findings
Chongqing, China (Zhou et al., 2020) [115]	Recognizing the cultural ecosystem services of two wetland parks	Used the social values for the ecosystem services (SolVES) model and preferential survey data	The high value of cultural ecosystem services of Xiuhu Wetland Park was transferred to enhance Guanyintang Wetland Park
Taipei, Taiwan (Chen et al., 2019) [58]	Investigating how to use quantitative and qualitative measure of wetlands to develop plans for the protection of the wetland's environmental resources.	Conducted an experimental study by using DEMATEL and DANP models	DEMATEL can be used with DANP to construct a new measurement model for the effects of wetlands environment plan
CMA, Chile (Rojas et al., 2019) [116]	Assessing the impacts of urban planning and development between 2004-2014 on wetland conservation	Used SEA as a framework and under a MUPC urban growth around the wetland planned and implemented	The inefficiency of wetland conservation under MUPC and this plan is not sustainable enough
Review paper (Roy-Basu et al., 2020) [111]	Suggesting a sustainable model	Adaptive co-management to prevent further wetland degradation	Contact relevant actors, build capacity, and implement protection measures
Ria Formosa, Portugal (Sousa et al., 2020) [117]	Evaluating the impact of economic activities on the process of wetland rehabilitation	A temporal and spatial perspective of land use change in wetland areas	The rehabilitation of the wetland in Ria Formosa since the late 19th century was a process dependent on the success of economic activities
Hunan, China (Luo et al., 2019) [118]	Evaluating the sustainability of an artificial wetland	Used a model of probabilistic linguistic preference relations	The sustainability of artificial wetlands plays an important role in the local environment, society, and economics development
Quebec, Canada (Varin et al., 2019) [119]	Developing a spatial analysis tool to quantify ecosystem services to help wetlands sustainable management	Used object-based and photo-interpretation approaches	The expansion or accumulation of many wetlands may decrease the biodiversity in wetlands

Table 4. Cont.

Thus, this paper examined the trends, reasons, extent, and effects of landscape conversion over the past few decades. The paper introduced a model called adaptive comanagement to prevent further wetland degradation. The adaptive co-management model refers to the convergence of two independent concepts of adaptive management and comanagement. For the long-term sustainability of the wetlands, adaptive co-management is integrated with five environmental, political, physical, economic, and institutional aspects.

In summary, this paper provided the following key recommendations for regulatory, institutional, development, and management authorities. The government should contact relevant actors to raise funds for research, capacity building, and implementation of conservation measures. Participation of the locals in decision-making and planning committees is necessary to ensure the inclusion of their invaluable knowledge in conservation action.

For strong awareness, acknowledgments are needed, and educational programs should be planned. In addition, civic institutions must work with the government to implement adaptive co-management and sustainability plans. The media should also use prominent programs with corporations to promote the cause of wetland protection.

Su et al. [112] identified the ecosystem value and sustainability of the Liaohe Estuary wetland (LEW). This study aimed to (i) define the wetland flows based on the sustainability analysis, (ii) assess the relationship between the input and output of the LEW, and (iii) investigate the sustainability of the wetland ecosystem and suggest a sustainable future for the LEW through the sustainability analysis.

For sustainability evaluation, this study used Emergy analysis. Emergy assessment is a method to estimate the capacity for sustainable development of regional eco-economic systems at different scales. For example, "Emergy converts energy or currency that cannot be directly compared in the natural environment and economic systems into the same magnitude index".

An ecosystem service function was used to value the area to the humans. The results showed that human activities, developments, and investments in the LEW area play a critical role in protecting the wetland ecosystem. Due to diverse human activities, the environmental loading ratio indicated high environmental pressure, and the LEW area became unsustainable. In summary, the study illustrated that the services offered by LEW have great ecosystem potentials. Wetlands not only facilitate environmental benefits for the societies but also provide urban sustainability.

Byeon and Nam [113] investigated the treatment of non-point source (NPS) pollution and biodiversity increases in a constructed wetland. NSP stands for the diffusion of pollutants from a wide area, compared to point source pollution. They used the sustainable structured wetland biotope (SSB) system with a short retention time and high biodiversity as a treatment wetland for non-point source pollution.

The SSB system is used to investigate site-specific environmental conditions and maximize water treatment efficiency. In this research, the SSB system included a Forebay area (primary retention basin), a micro pool area (secondary retention basin), and multi-cell structures of marshes and ponds. For evaluation, a wetland and a rainwater detention tank were constructed with the SSB system to treat NPS pollution downstream from Jecheon city, located in the upper river basin of the Han River in the Republic of Korea.

In addition, the filter media and the retention basin were partially compensated for the seasonal change of the treatment. The overall flow of water was also maintained at a low level. As a result, the SSB wetlands indicated a treatment impact on NPS pollution at the initial rainfall events within a moderately short retention time. The research also suggested that this methodology can control urban NPS pollution while providing access to an area with high biodiversity for residents in urban areas with drastic land use.

A study by Wu et al. [114] addressed the effects of wetland rehabilitation on a coastal environment. The research considered two objectives. First, they aimed to identify detailed land use and vegetation changes due to wetland rehabilitation. Second, they sought to determine how to balance land rehabilitation and ecological protection in coastal areas. Time-series analysis of images illustrates that from 1989 to 2013, ca. 9793.4 ha of coastal wetlands were separated from the sea and enclosed to inland wetlands.

The separation interrupted the exchange of sediment and water flux between wetlands and the coastal ocean. The results indicate that the coastal wetland ecosystem structure was destroyed or replaced with a drained and impervious surface with increased human activities. This means that the main elements of the wetlands, including water, soil, and vegetation, were destroyed. In summary, considering the resilience of the wetland ecosystem, a balance should be established between land use and wetland rehabilitation to reduce the conflict between economic growth and coastal ecological security.

A study by Zhou et al. [115] aimed to recognize the cultural ecosystem services of two wetland parks. These parks were the Xiuhu Wetland National Park and Guanyintang Urban Wetland National Park. The study used the social values for ecosystem services (SolVES) model and preferential survey data. The SolVES model (http://solves.cr.usgs.gov, accessed on 15 September 2021) a GIS application, was developed to assess, map, and quantify the social values of ecosystem services. The SolVES application is suitable for evaluating small-scale waterbodies (e.g., wetlands and ponds). In Xiuhu Wetland National Park, the cultural ecosystem services were graded.

The services included the perceived biodiversity, future and historical value, the value of beauty and recreation, and the value of culture and learning, as well as spiritual and therapeutic values. From the gender perspective, while men preferred aesthetic, cultural, recreational, spiritual, and therapeutic values, women were primarily interested

in historical values. From these analyses, the value of cultural ecosystem services of Xiuhu Wetland National Park was transferred to Guanyintang Urban Wetland National Park. As a result, the Guanyintang park has performed well from the cultural ecosystem services point of view.

Chen et al. [58] investigated how to use quantitative and qualitative measurements of wetlands to develop plans for the protection of a wetland's environmental resources. The research also presented a model to solve real problems and an experimental study to achieve sustainable development at the desired level. As a methodology, the research utilized a hybrid Multiple Attribute Decision-Making (MADM) model and a decisionmaking trial and evaluation laboratory (DEMATEL) technique. The technique was used to identify complex relationships to create an effective network relation map.

The method was an experimental test to assess and measure the actual conditions for improving the environmental problems of the wetland. The technique also enhances the effectiveness of wetland resource plans and improves the strategic target plans. The aim is to achieve the interest levels for human welfare and sustainable development.

Regarding the findings, the research presented a combined model using DEMATEL and DANP models. DEMATEL can be used with DANP to construct a new measurement model for the effects of wetland environmental planning. It is suitable for complicated decision-making problems involving environmental planning, psychology, and human resources management. In short, the model is seen as relevant and suggested for use, for instance, in areas of soil and water conservation, ecosystems, environmental development, and landscape and rural development.

A study by Sousa et al. [117] evaluated the impact of economic activities on the process of wetland rehabilitation. This study considered a temporal and spatial perspective of landuse change in wetland areas. They examined the feasibility of productive activities, such as aquaculture activities, without compromising the natural system while creating longterm sustainable economic development. The study applied a combination of mapping and documentary data to assess the wetland change in the Ria Formosa as a function of economic activity dynamics. They also integrated various datasets into a GIS and tracked the wetland land-use change over the past 130 years at intervals of a decade.

The results indicated that the rehabilitation of the wetland in Ria Formosa since the late 19th century was a process dependent on the success of economic activities. The historical analysis of wetland land-use changes also proves the importance of sustainable economic management for the study area. The recession periods led to the abandonment of wetland's revitalized areas. As a solution for developing new productive units, the research proposed adopting an ecosystem approach to aquaculture as a management strategy by creating an aquaculture management area. The management strategy can be used to stimulate small-scale aquaculture through the networking of more profitable production systems with reduced environmental impacts.

Furthermore, small-scale aquaculture units can improve market development and supply chain sustainability in sensitive environmental areas, such as coastal wetlands. Aquaculture ponds can help create wildlife migration, food supply, and natural corridors used by waterbirds. Bird watching areas can also have many benefits and ultimately contribute to the growth of ecotourism in the area.

Research by Luo et al. [118] evaluated the sustainability of an artificial wetland using a model of probabilistic linguistic preference relations (PLPRs). Since the decision-making process involves uncertainty and fuzziness, the decision-makers (DMs) often have difficulty expressing their opinions. Thus, this research aimed to study the impact of decision-makers' decisions on the sustainability of artificial wetlands. The study used the group decisionmaking (GDM) approach to assess the sustainability of constructed wetlands under PLPRs.

Hence, the research adopted linguistic preferences information, a straightforward procedure using information directly from the DMs. Therefore, linguistic preferences can reduce the cost of inaccuracy to some extent. The research formulated a mathematical model based on the trust degree for determining the weight of DMs in the GDM. The trust degree defines the trust relationship among DMs. The research proposed a cosine similarity measure called probabilistic linguistic term sets (PLTSs), which is known to be a good choice for information expression.

In most cases, PLTSs accurately and clearly describe the DM preferences. Thus, the mathematical model enables selecting the best-constructed wetland in Hunan province. In summary, the results showed that the sustainability of artificial wetlands plays a vital role in the coordinated development of the local environment, society, and economics.

Rojas et al. [116] evaluated the impacts of urban planning and development on wetland conservation between the years 2004 and 2014. They assessed the urban development and its impacts on pre-preserved areas, reducing flood risks, and projected under the Metropolitan Urban Plan of Concepción (MUPC). They also discuss the consequences of land zoning in coastal regions to provisioning ecosystem services and flood controlling against frequent tsunamis on the central coast of Chile. This study used a remote sensing method to detect the wetland area and the remaining Land Use and Land Cover (LULC) categories for 2004 and 2014.

Remote sensing was used to select the most abundant wetland vegetation to improve its detection in contrast to the surrounding LULC categories. The study considered the whole area of the wetland and its surroundings with a distance of approximately 300 m. Then, the study used two Landsat images and one Landsat 5 TM (Thematic Mapper) for 2004. A Landsat 8 OLI/TIRS (Operational Land Imager/Thermal Infrared Sensor) was also used for 2014.

The study also used maximum likelihood classification (ML) for images—a supervised classification widely used for Landsat satellite imagery. Then, 25 training areas, i.e., control points, were selected for each classification category, a total of 200 categories, to evaluate the images. These categories included the built-up, water bodies, mixed forest, planted forest, scrublands, grasslands, beaches and dunes, and wetlands.

In the methodology, first, the relevant GIS layers were combined with the geographical form map of the wetland. Then, the impact of both new and planned urban development was investigated on the geographical form of the wetland. The effect of both new and planned urban development was investigated on wetland loss. In addition, the effect of the tsunami was examined on these urban areas.

As a result, this study showed the unsustainability and inefficiency of wetland conservation under MUPC. This is because the wetland areas of MUPC were significantly decreased to about 60% of their current size. It also appeared that both new and planned urban development areas under the MUPC are in danger of flooding caused by a tsunami. In addition, strategic environmental assessment is needed to reduce the adverse effects of urban development and contribute to the sustainability of the study area.

Varin et al. [119] developed a spatial analysis tool to quantify ecosystem services (ES) to help wetlands sustainable management decisions at the watershed scale. This research aimed to:

- (*i*) recognize and design spatial indexes to represent the functions of ecosystem services;
- (ii) extend an approach with two spatial scales, one for strategic planning and the other for local interventions;
- *(iii)* map a complete watershed using the past, present, and future wetland ecosystem services; and
- (*iv*) ensure that the approach is usable for other watersheds in the same area and other ecosystem services.

The research used several methods for preparing spatial data, i.e., maps. For instance, the object-based and photo-interpretation approaches generated recent and historical landuse data for 2011 and 1984, respectively. Therefore, Landsat TM satellite imagery with a spatial resolution of 30 m was used to prepare the maps. Then, these maps were used to simulate future land use for the year 2050. The study also considered two wetland management scenarios for comparison between 1984, 2011, and 2050. The scenarios were an optimistic scenario between 1984 and 2050 and a pessimistic scenario between 2011 and 2050.

These were used to evaluate ecosystem services over a historical period of 27 years (1984–2011) and a forecast period of 39 years (2011–2050). Furthermore, the ecosystem services quantification method was used to describe the functions of the wetlands. To prepare the ecosystem services indicator (ESI), the technique summarizes the weight and standardized function indicators (FI). The ESI describes the benefits to the community and consists of three types of FI, including ecological, socio-cultural, and economic indicators.

Based on sustainable development, these indicators are utilized to measure the conserving habitat support services' efficacy spatially. Six function indicators were selected to evaluate ecosystem services, four of which are the ecological FIs, and the other two are the socio-cultural FIs. These FIs include wetland connectivity, habitat fragmentation, water connectivity degree, anthropogenic barrier, natural heterogeneity, and natural aesthetics.

The results of the two wetland management scenarios, i.e., the optimistic and the pessimistic scenarios, showed that, in 2011, there were more critical areas than in 1984. This was due to the significant loss of wetlands in the territory between these two times. In the pessimistic scenario between 2011 and 2050, many sub-watersheds are destroyed; however, these sub-watersheds can be improved in some cases. The improvement can be made by fragmenting some large wetlands into several smaller wetlands.

This can be potentially beneficial for wetland connectivity and natural heterogeneity. In the optimistic scenario between 1984 and 2050, some sub-watersheds are worse than in 2011. The expansion or accumulation of many wetlands may decrease the diversity of the wetlands by reducing the connectivity. This illustrates the significance of the choice of FIs and how a wetland can develop.

To conclude the section, urban wetlands contribute to urban sustainability by providing ecosystem services to human societies. However, human activities, urban developments, and the changes in land use in urban areas affect the sustainability of wetlands, leading to challenges in urban sustainability. The studies showed that where human activities increase, the ecosystem structure of wetlands is destroyed. This results in the decrease of wetland efficiency and, thus, reduces urban sustainability. Therefore, to reduce the conflict between increasing human activities and reducing the ecological efficiency of the wetlands, studies suggest a balance between land use and wetland restoration. Indeed, the development of urban areas without considering protecting natural resources, such as wetlands will emasculate urban sustainability.

Unfortunately, the ecosystem services provided by wetlands are often undervalued and even ignored during the planning and design stages of urban development. Therefore, it is essential to inform governments and increase public awareness and stakeholders about the importance of urban wetlands to consider them as a nature-based solution in urban and peri-urban areas. This information is indeed mandatory to plan future sustainable cities.

6. Urban Wetlands and Recreational Values

Wetlands provide significant opportunities for nature-based recreation [16]. The economic benefits of wetland recreational services can be used as a basis for the protection of wetlands. The extensive recreational opportunities created by the wetlands include fishing, swimming, wildlife viewing, and hiking [120]. People who visit wetlands for recreational benefits perceive and highly value these nature-based resources, especially since other kinds of recreation cannot easily replace these activities. [31,121,122]. Consequently, recreational benefits can provide a stronger motivation for people to maintain wetlands ecosystems [123,124].

To highlight the recreational values of the urban wetlands and understand the methodologies and approaches for calculating these values, in the following, we review some existing studies in the literature and summarize the objectives, methods, and findings of the reviewed literature articles in Table 5.

Reference	Objective	Methodology	Findings
Valencia, Spain (Vidal Gimenez and Ruiz Mas, 2020) [125]	Analyzing the economic values of the recreational services for wetlands	Used the travel cost method (TCM) and the contingent valuation methods (CVM).	The economic crisis had "no direct impacts" on the economic valuation of the recreational aspects of these wetlands
Muzuma, Tanzania (Musamba et al., 2012) [126]	Evaluating the recreational value of Lake Victoria in Musoma Municipality	Used a questionnaire to a sample of 120 recreationists	A limited number of recreational activities around the Lake Victoria
Kerala, India (Sinclair et al., 2018) [127]	Mapping nature-based recreation patterns, and investigating how recreational benefits are affected by changes in ecosystem quality	Used an economic valuation tool for recreational ES. Used a crowd sourced travel cost method and geotagged photographs	The results of the CTCM are consistent with economic theory for both the travel cost and income variables
Costa Brava, Spain (Pueyo-Ros et al., 2018) [128]	Evaluating the socioeconomic value of the ecological restoration of a coastal wetland on Costa Brava (Spain)	Developed a model combination of travel costs and contingent behavior to assess restoration on the recreational value	The results of the TC + CB model indicated that wetland recreational value was not significantly affected by restoration
Finland (Lankia et al., 2019) [129]	Investigating the impact of water quality changes on swimming behavior and recreation benefits.	Conducted a survey include two types of questions (1) general information, such as age and gender; (2) water quality perceptions of water clarity	Reduction in water quality to a level where water visibility is less than 1 m was associated with the recreation value being reduced
Finland (Vesterinen et al., 2010) [130]	Investigating the association between recreational participation for water activities and the water quality	Used the national recreation inventory data combined with water quality data to model recreation participation	Water quality had no effect on boating. Improving the water clarity increased the frequency of swimming and fishing
Finland (Lankia et al., 2020) [131]	Conducting a national-level accounting study for nature-based recreation in Finland	Recreation ecosystem services was defined as the number of recreational day visits to nature areas	The location of recreational areas is one of the main elements of recreational opportunities
Zarivar, Iran (Aazami and Shanazi, 2020) [132]	Exploring livelihood effects of wetlands on people's livelihood	Used a quantity–quality approach conducting library and field studies along with questionnaire and a focus group discussion	The wetland had a great effect on residence's life in five dimensions of livelihood capitals, i.e., financial, natural, human, physical, and social
Iowa, US (Park et al., 2017) [133]	Exploring visitors' perceived benefits of wetlands and examining residents and non-residents differ	Developed an extended model of goal-directed behavior to examine the impact of perceived benefits of wetlands	Residents showed that a positive attitude toward wetlands played a significant role in forming a strong desire to visit wetlands
Denmark (Odgaard et al. <i>,</i> 2017) [45]	Analyzing which ecosystem services indicators defined the placement of wetlands and what are the recommendations for future selection for potential wetland reconstruction	Used the multi ecosystem service value-driven method to drive the optimal placement of wetlands in terms of maximizing selected ecosystem services that wetlands can provide	35 identified catchments with potentially high suitability, recreation potential, high biodiversity, and low land rent have not been prioritized

Table 5. Urban wetlands and recreational values.

Vidal Gimenez and Ruiz Mas [125] investigated the impact of the economic crisis on the economic valuation given by visitors to three natural parks with wetlands. The research compared the results of a questionnaire collected in 2004–2005 as favorable years of economic conditions and a survey conducted in 2013–2014 as unfavorable conditions. Therefore, the research applied the travel cost and the contingent valuation methods to offer an approach to the economic values of the recreational services for the three wetlands.

The results based on a comparison did not show a clear impact with different behaviors among natural areas between the values of the two time periods, which had a decade difference. The research stated that it can also not be concluded that the economic crisis directly impacted the economic valuation of the recreational aspects of the wetlands. In some parks, the situation led to a lower valuation, but the crisis led to higher valuation in others. In all cases, visitors evaluated the recreational aspects of the parks positively. However, this evaluation was not related to the economic factors but the social awareness of the environment.

Musamba et al. [126] addressed the recreational value of Lake by applying the travel cost method. Lake Victoria offers a wide range of recreational activities, including fishing, bird watching, a lake view site, boating, and camping activities. This study used primary and secondary data. The primary data included the background of wetlands, land uses, participants' socio-economic characteristics, and recreational activities of Lake Victoria. In contrast, the secondary data were collected by interviewing 120 local people and visitors. Attention was paid to the number of visits, visit seasons, expenditures, and visited sites in the collected information.

People were also asked to indicate their willingness to pay for goods and services provided by Lake Victoria to improve its quality. The willingness to pay was considered a good reflection of the value of the wetland. Then, the collected data was investigated using the Statistical Package for Social Sciences (SPSS) software. The quantification of wetland benefits was also performed using Microsoft Excel. The study showed that Lake Victoria has a high capacity to support the livelihood of local people. However, there was a limited range of recreational opportunities, but the recreational value of Lake Victoria contributes significantly to the income of local people.

Sinclair et al. [127] introduced an economic valuation tool for recreational ecosystem services that is suitable for large spatial scales. This tool was based on crowdsourced metadata, i.e., geotagged photos collected from social media sources in Vembanad Lake. This study used the metadata of 47,246 geotagged images from the Flickr website, retrieved using Flickr's Application Programming Interface (API) between 1 January 2005 and 31 December 2016. The geotagged metadata is used as a proxy for human behavior. The metadata includes the user identification number, location, and timestamp of the time of the shooting. Therefore, the study used the retrieved metadata to:

- (i) map nature-based recreational patterns,
- (*ii*) identify the value of recreational ecosystem services, and
- (*iii*) examine how recreational benefits are affected by changes in ecosystem quality.

This study mapped the geotagged photos on the Vembanad Lake and used GIS to identify the recreational hotspots. Then, the Flickr images were analyzed to predict the home locations of the users who did not share their information in their public profile. Next, geotagged photos were used to make a single site "crowdsourced" travel cost method (CTCM) of the Vembanad lake. The consumer surplus per trip by combining visitor frequencies with inferred locations for visitors was also achieved.

This technique provides a tool for the economic valuation of recreation at large scales and low implementation costs. The study further combined the estimated per-trip benefits with the visitation rates under different water quality and wetland extension scenarios to investigate the implications of different management approaches. The study concluded that crowdsourcing methodologies, such as utilizing geotagged metadata from social media, were efficient approaches for valuating recreational ecosystem services.

In addition, the results of CTCM, which represents the mapping of nature-based recreational patterns, showed that the results were consistent with economic theory in terms of travel cost and income variables. Therefore, the CTCM is a suitable technique for estimating the information of individuals who do not voluntarily share it in their public

profiles. Moreover, the study showed that improving water quality to the extent that it supports wildlife and fisheries would increase the economic benefits (almost 4 million USD) annually from the wetland's recreational services.

Pueyo-Ros et al. [128] assessed the socio-economic of a coastal wetland located of tourist destination named Pletera. The study analyzed whether the ecological restoration of the coastal wetland would affect its recreational demand. The study consists of a survey using a questionnaire from the visitors and residents of Pletera from July to September 2015, which are the peak visit months. A combined model of travel costs (TC) and contingent behaviors (CB) was used to evaluate the impact of ecological restoration on the destination's recreational value. After obtaining the visitor profile, a cluster analysis and subsequent comparisons were performed to understand how restoration affects tourists' contingent behavior.

The visitors were also divided into three groups, including indifference, recreation, and protection. The indifference group, which included 109 visitors, had the lowest ecological preferences of Pletera. The recreation group with 42 visitors was characterized by three variables: access to the wetland by car, the access restrictions, and the visitors' gender, i.e., 100% male. The protection group had 81 visitors who had more visits than the other two groups; they used cars the least compared to the other groups. This group also had a more eco-centric perspective.

The results of the combined model of TC and CB depicted that the restoration did not greatly influence the recreational value of the wetland. The cluster analysis of the three groups, i.e., indifference, recreation, and protection, illustrated three different attitudes toward the ecological restoration of the wetland. The results showed that the visitor profile highlights the difference between the actual and contingent visit rates.

Visitors attracted by the natural environment visit the wetland more often, while visitors who use the wetland for recreational purposes usually have fewer visits. The methodology used in the study showed that the changes in recreational demand after ecological restoration highly depended on the visitors' sociodemographic profiles. The results also suggested that ecological restoration will favor visitors and the tourism sector and positively impact the Pletera region's demand, including recreational beach opportunities.

Lankia et al. [129] investigated the impact of water quality changes on swimming behavior and recreation benefits. The core objective of this paper centered on swimming since swimming is the most common water recreation activity in Finland. This is an activity that is directly affected by water quality. Thus, the paper conducted a survey in the fall of 2009 using the internet and e-mail. A study of a total of 1644 sent e-mails achieved a response rate of 41.4%. The survey included two types of questions, including:

- general information, such as age, gender, place of residence in Finland, distance to typical visiting swimming sites, number of visits per year, and travel costs;
- (2) water quality perceptions included questions about water clarity, i.e., water depth and visibility. In addition, participants were asked whether there were sludge and rocks in the water.

Applying the combined travel cost and contingent behavior method showed that the recreation value of a swimming trip with the current water quality was approximately 16 euros per trip. The results showed that the hypothetical reduction in water visibility to less than 1 m, and a large amount of sludge reduced the recreational value to 9 euros per trip. Improving water quality such that the water visibility is more than 2 m and if there is no increase in sludge, the recreation value of each trip will be 22 euros. In addition, the total annual recreation value of all swimming visits when improving the water quality enhances the recreational benefits by 80–53%. In contradiction, declining the water quality reduces the benefits by approximately 80%.

Vesterinen et al. [130] investigated the association between recreational participation in water activities and water quality. The water activities included swimming, fishing, and boating, and water clarity was used as an indicator of water quality. They also evaluated the consumer surplus of a water recreation day and the marginal social net benefits of an exogenous improvement in water quality. This study first modeled the water recreation participation and participation frequencies for each water activity. Then, they applied the TC method to estimate the value of one water recreation trip.

The value of a water recreation trip was assessed using the TC method for the three recreational activities. The technique also estimated the annual frequency of a one-day trip from home to the last-visited water recreation site. The results showed that water quality did not affect boating, but improving water clarity increased the swimming frequency and the number of anglers and fishing near home waters. The distance to the nearest recreational water did not prevent participation in fishing or boating.

An association between socioeconomic variables and water recreation participation was also found. The number of annual swimming trips and the number of anglers increased with the water clarity. In the water policy scenario, improvement of the water clarity up to one meter for inland and coastal waters was associated with a consumer surplus increase of 6% for swimmers and 15% for fishers.

Lankia et al. [131] conducted a national-level accounting study for nature-based recreation in open access conditions using the United Nations System of Environmental Accounting–Experimental Ecosystem Accounts (SEEA-EEA). This research investigated how recreational services could be integrated into an ecosystem accounting framework at the national level as one of the cultural ecosystem services. The study suggested a model describing how this type of accounting is constructed and how different national data sources assist recreational accounting in physical and monetary terms.

The research calculated the monetary value of outdoor recreation based on consumer surplus estimates achieved applying the travel cost method. As a result, there are almost unlimited possibilities in Finland to enjoy natural areas because of the right of public access for recreational purposes. The area of recreational ecosystems includes 25.8 million hectares of forest land available to the public for recreational purposes. The natural areas for aquatic recreation include 3.4 million hectares of inland water resources and 5.2 million hectares of marine waters.

Based on the research, natural areas in Finland, such as forests, parks, water areas, and, e.g., wetlands, are used for recreation. A total of 90% of recreational visits close to home (daily visits) occur at least in part in forest environments. A total of 50% of the visits occur in parks and open spaces, and 75% include aquatic environments, lakes, ponds, rivers, streams, and marine and coastal environments. Moreover, the location of recreational areas is one of the main elements of recreational opportunities. The presence of natural areas with high recreational facilities for daily use offers well-being to people, particularly in urban areas.

Aazami and Shanazi [132] investigated the impacts of wetlands on people's livelihood through wetland capacity, mainly from tourism and agricultural-related activities in Iran. In this study, two qualitative and quantitative research methods were used. A questionnaire consisted of three sections: A, B, and C. Section A included information about the demographic characteristics of the respondents, such as gender, age, marital status, occupation, and household size.

Section B had information on livelihood assets and wetland community strategies, such as human, social, financial, environmental, and physical assets, and livelihood strategies to support their income. Finally, Section C included information on the vulnerability of wetland community development. The field of vulnerability was divided into three types, namely shocks, trends, and seasonality. Shocks refer to some unexpected occurrences that might affect community livelihoods. Trends refer to changes over time in natural resource stocks and quality that impact community livelihood. Seasonality refers to seasonal changes that constrain the livelihood choices of people.

The results showed that households have different strategies for responding to vulnerabilities. The sale, consumption, and use of wetland goods and services and improved assets were their most important protection against natural and economic shocks. The study indicated that the wetland has contributed to the profitability of small family businesses (tourist services, handicrafts, retail, etc.) and has increased agricultural income and thus their livelihood. Moreover, natural capital was the strongest asset for measuring people's sustainability, having a positive impact on their livelihood. The potential of the wetland in agriculture and livestock, fisheries, and tourism led to improved production and income, stability, and sustainability of the economy. It also provided livelihoods and job opportunities in the region.

Park et al. [133] developed an extended model of goal-directed behavior to analyze the impact of perceived benefits of wetlands on the decision-making process and identified differences between residents and non-residents. The objectives of this study were to:

- (*i*) examine whether the perceived benefits of wetland visitors were important in influencing their desire and future intention to visit wetlands,
- (ii) investigate whether residents and non-residents differed in the causal relationships depicted in the destination decision-making process model.

As a methodology, an online survey was conducted regarding outdoor recreation in Iowa. Out of the 670 people who took part in the survey, 462 were valid for data analysis. A theoretical framework was created as an extended MGB (EMGB) by incorporating the perceived benefits of wetlands into the MGB. The EMGB was recognized as an updated approach to the TPB and MGB to better explain the decision-making process based on multiple cognitive and affective components. The study used an extensive model of goal-directed behavior (MGB) to investigate the decision-making process.

This model describes the decision-making mechanism concerning various factors, such as desire, past experiences, motivational and emotional factors, attitude, subjective norm, and perceived behavioral control, which are introduced in the theory of planned behavior (TPB) and the theory of reasoned action (TRA).

The model of structural equations showed that the perceived benefits of wetlands influence the decision-making process. Desire as a key element (conceptually different from behavioral intention) was essential to link attitudes, predicted emotions, and the frequency of past behavior with behavioral intent. Residents, compared to non-residents, showed that positive attitude and past experience influenced the desire and intention to visit wetlands. The study revealed that the perceived benefits of wetlands played a vital role in strengthening the decision-making process for visiting wetlands.

In addition, marketing managers need to consider the differences between residents and non-residents in the decision-making process for advertising or effective advertising. Moreover, residents showed that a positive attitude towards wetlands plays an important role in creating a strong desire to visit wetlands. They were more likely to revisit wetlands if they were frequent visitors in the past, compared to non-residents. Therefore, tourism operators are encouraged to identify frequent visitors as an important market among residents and develop an appropriate marketing plan to meet the needs and demands of consumers.

Odgaard et al. [45] introduced a new multi-criteria method based on ecosystem services value to optimize the location of restored wetlands to maximize the selected ecosystem services that a wetland can provide. First, five ecosystem service indicators were mapped for wetlands in Denmark (recreational potential, biodiversity, nitrogen mitigation potential, inverse land rent, and flash-flood risk).

For this purpose, the analysis of the catchments that accommodate a reconstructed wetland was performed. Then, the scenario test and analysis of the hotspots were combined to provide future recommendations for the optimal placement of wetlands. The studied scenarios were climate adaptation, aquatic environment protection, land-based economy, and rich nature. Based on these scenarios, the most suitable areas for wetland reconstruction were mapped considering the scenarios and weights of ecosystem service indicators.

The results showed that current reconstructed wetlands were situated in catchments with higher land rent (i.e., agriculture intensive areas), higher nitrogen mitigation, and to some extent, higher flash flood risk. In contrast, lower biodiversity compared to catchments without reconstructed wetlands. Therefore, recreational capacities, increasing biodiversity, and low land rent are not a priority. When optimizing all scenarios, 35 of the 3023 catchments surveyed were particularly suitable. However, out of the 35 identified catchments with high potential, only two hold a reconstructed wetland. This represents a previous placement almost without considering the maximum benefits of ecosystem services.

To conclude this section, there are different forms of wildlife species in urban wetlands that make wetlands attractive places for recreation and leisure time. Indeed, wetlands provide countless recreational activities for people, such as walking, boating, swimming, fishing, trapping, and bird watching. The recreational opportunities offered by water bodies contribute to both physical and mental health. The studies demonstrated that the number of water areas, population, age ranges of people, water quality, and the type of recreational activities were linked in urban areas. The higher the population was, the higher the recreation activity. The water quality also defined the swimming rate, and the age range specified the type of activity, such as swimming, fishing, and boating. Younger people were interested in swimming, and older people were interested in finishing and boating.

7. Urban Wetlands and Social Perceptions

Considering the social perception of natural ecosystems is crucial to the success of conservation programs [134]. The central aspect of this social and environmental perception goes back to the use and non-use of various values that people attribute to natural areas. These are referred to as social values [135]. Successful conservation of natural ecosystems occurs when we identify ecological and economic priorities for specific areas [136]. It also depends on how these priorities align with the social values allocated to these areas by the community [134,137].

Thus, considering the different ways human beings perceive and value natural areas, the social acceptance of conservation plans can be increased [138], and the public can better comprehend the complexity of conservation values [139]. Comprehension of the connections between ecological and social values can help specialists design strategies, e.g., information exchange, incentive schemes, for particular areas that are effective and efficient [140].

Ecosystem assessment has commonly been carried out in a purely ecological framework, regardless of social functioning. However, more recently, it has been determined that we must also define some social measures to preserve and restore the ecosystem [141]. Naturally, everyone interacts with ecosystems and landscapes and makes value assessments. These value assessments reflect how they think and react to these environments [142]. Therefore, people are sensitive to how ecosystems function, landscape features, and change over time [143]. However, the perception of the ecological situation and value estimated by experts may conflict with public opinion [144].

Therefore, these differences must be adapted. The adaptation can be made by linking the value and function of the ecosystem, i.e., by identifying the ecological parameters that adequately describe the ecological function and perceived value both for people and experts [141].

Different values may be related to the environment. These values include aesthetics, recreation, biology, and economics [145]. For example, perceiving the aesthetics of the environment is directly related to emotional processes and may create a strong social motivation to preserve ecosystems, without considering environmental function or vulnerability. Therefore, considering aesthetic perception is useful for environmental protection and restoration because it increases the motivation of social support for environmental protection [146].

People's perceptions and preferences about ecosystems are impressed by personal, geographical, and social characteristics. These characteristics include age, gender, income, political orientation, environmental organizations, moral beliefs, use and non-use of specific areas, life experiences, and living environment [147].

Due to the importance of improving social perceptions about the urban wetland and understanding the possible approaches, in the following, we conducted a review on this topic, and we summarize the objectives, methodologies, and findings of the reviewed articles in Table 6.

Reference	Objective	Methodology	Findings
South Korea (Do and Kim, 2020) [148]	Analyzing metadata and contents of the photos taken in South Korean wetlands	Using API, respectively 69 and 287 geotagged photos collected from wetlands	Preferred visiting times depended on tourism attractions in specific seasons as well as parts of the day
New York, US (Sonti et al., 2020) [149]	Examining how and why people use the "natural" and "landscaped areas" of New York city parkland	21 parks were selected in five New York City boroughs, including parks larger than 400 acres	More people visit landscape areas to bring children to the park and visit forest areas and wetlands much less than the landscape areas
Victoria, Australia (Dobbie, 2013) [150]	Investigating the public aesthetic preferences and its relationship with conceptual perceptual tools for wetlands	Used selected connotative perceptual constructs that were rated on a 7-point Likert scale	Proved preference for water and vegetation is relevant aesthetic feature. The preference of the wetland improves with increasing amount of water
Benslimane, Morocco (Bouahim et al., 2015) [151]	Investigating the social perception of ecosystem services and vulnerability analysis associated with temporary-flooded wetlands	101 ponds were selected and conducted interviews with well-known experts and 110 household	60% of respondents found ponds very useful and highlighted place attachment and cultural values
Tubul-Raqui, Chile (Rojas et al., 2017) [147]	Evaluating the social perception of ES in a coastal wetland after the earthquake in Tubul-Raqui, Chile	Used surveys and data collection to analyze the social perception or perceived benefits of ES	The highest value is related to cultural services. Wetlands can provide opportunities for services related to recreation, local identity, science, and environmental education
Lyon, France (Cottet et al., 2013) [141]	Investigating the relationship between perception of wetland aesthetics and healthiness with wetlands ecological functioning	A questionnaire containing photos of the wetland was presented to people who were asked to evaluate the two perceptual criteria of aesthetics and environmental health	People's perceptions of the aesthetics of the wetland strongly depended on visual criteria
Xochimilco, Mexico (Torres-Lima et al., 2018) [152]	Studying the variables of socio-economic, environmental, and regional economic management that affect the population's perceptions of wetland	Used an approach that focuses on area characteristics and fieldwork was conducted between April and May 2010	Education helps to create a positive local perception of urban wetland. For some participants value of wetland based on the aesthetic qualities and its potential use as a reserve for human habitation
Nova Scotia, Canada (Manuel, 2003) [153]	Investigating public awareness and perceptions of small urban wetlands in Halifax and Nova Scotia, Canada	Three wetlands were selected for examination and use of questionnaire with 65 participants	Participants were interested in the environment and recreation. Residents liked wetlands and had positive views about them

 Table 6. Urban wetlands and social perceptions.

Reference	Objective	Methodology	Findings
Alberta, Canada (Rooney et al., 2015) [55]	Comparing NWs with CWs to examine residents' perceptions of the environmental services of NWs and CWs with their biophysical values	Three sample sets of 20–34 people were considered for interview to conducted a survey of social values	More than 80% of participants reported that NWs value biodiversity and groundwater recharge services, nearly 90% of the participants found value in the aesthetics of the CWs
Helsingborg city of Sweden (Pedersen et al., 2019) [154]	Identifying urban wetland areas contribute to the well-being and quality of life	Environmental psychology tools were distributed in questionnaires in three municipalities with wetland areas	Wetland areas were perceived by residents as contributing to their quality of life

Table 6. Cont.

Sonti et al. [149] examined how and why people use the "natural" and "landscaped areas" of New York City parkland. They used random field interviews with 955 park users to examine the differences in park use and the reasons for visiting the park. In this interview, when looking at the differences in park use and the reasons for visiting it, the site type and the gender of the respondents were considered. The data was collected from June to August 2014 to assess the use, value, and social significance of City parks that contain "natural areas."

As a case study, 21 parks were selected in five New York City boroughs, including parks larger than 400 acres and small parks containing at least one natural area. This research showed that more people visited landscape areas to bring children to the park and visited forest areas and wetlands much less than the landscape areas. For visitors to urban forests and wetlands, the feeling of shelter, attachment to the place, and the opportunity to experience nature were more valued.

At the same time, those who visit landscape areas were interested in a particular park quality or activity. Notably, urban wetlands mostly attracted walkers, dog walkers, and cyclists. Visitors who mentioned their priorities for landscaped areas were concerned that these natural urban areas are not safe and accessible to themselves or their children. The research also found that men and women perceived urban natural areas differently. Women were more likely to visit the park with their children and visit more landscapes than natural areas.

Moreover, people who participated in environmental groups visit the park's natural areas more often than those not involved in these groups. These results provide a platform for natural resource managers and urban planners as they investigate to improve park accessibility, visitor experience, and the perception of safety for all park users.

Dobbie [150] investigated the public aesthetic preferences and their relationship with conceptual perceptual tools for freshwater wetlands. For the sustainable management of wetlands, social preferences and sociodemographic attributes for wetlands in the urban area must be identified and understood. As a methodology, the study used selected connotative perceptual constructs rated on a 7-point Likert scale. The grades of evaluation were labeled "strongly", "moderately", and "slightly" with "neutral" describing the midpoint.

In the perceptual classification of Victorian freshwater wetlands, connotative constructs with the highest frequency of occurrence were selected to construct rating scales. Using the Kasmar environmental lexicon descriptors, these constructs were defined as 'orderly', 'open', 'healthy', 'natural', 'attractive', and 'varied'. Since perception primarily occurs visually, photos (N = 70) were used as a substitute for real landscapes in the perception of landscape and preference studies. Images were displayed to participants in two versions of the PowerPoint presentation with reverse slide order. Participants, including 241 adults (Minimum age 18), were selected from community groups in Melbourne, the capital of Victoria. Descriptive statistical analysis was performed using SPSS 14 software. Preference categories included five preference categories: 'brown grasslands', 'green grasslands', 'wetlands with emergent vegetation', 'treed wetlands', and 'wetlands with open water'. Different preferences in the wetland category were influenced by the presence of water and the type and color of vegetation. Statistically, there were three different groups: grasslands, open wetlands, and treed wetlands. Wet wetlands were preferred to dry wetlands.

"Green grasslands" and "brown grasslands" were equally least preferred, as 'grasslands', followed by 'wetlands with emergent vegetation' and 'wetlands with open water', as "open wetlands". 'Treed wetlands' were the most preferred. This study showed that, for wetlands, the preference for water and vegetation was a relevant aesthetic feature. The preference of the wetland improves with the increasing amount of water. In addition, it showed the effect of increasing the amount of water and green space on the preference of built and natural environments.

Preference for urban environments increases with the presence of water. As long as the edge between the land and water is visible, indicating safety when moving around areas, the preference for natural environments increased with water. The presence of water is an important sign of wetland health in wet wetlands, deduced from the clarity and movement of water and the existence of water vegetation. In dry wetlands, the color of the vegetation indicates health. In general, the color, and the freshness or dominance of green vegetation, increase the preference for natural landscapes.

Do and Kim [148] analyzed the metadata and contents of the photographs taken in South Korean wetlands. They paid attention to the color embedded in each object in the environment since the color is a stimulus that impacts the human mind and reflects the person's inner state. Therefore, it is assumed that the color of the wetland photograph shows the preferences of the wetland feature. Using the Flickr programming interface (API), the scientists collected 69 and 287 geotagged photos in Oppo wetlands and Suncheon bay.

Photographs were taken between 1 January 2007 and 31 December 2018. In this study, deep learning was used to analyze the images. Using Deep Learning Object detection, photographic subjects were identified from the contents of the image, which allows for quantitative analysis of the object. The results showed that many people were photographed near the starting points of wetland routes. Preference of visiting time depends on the tourist attractions in particular seasons and parts of the day. Investigation of the photos indicated that most visitors took photos of landscapes, such as fog and sunsets.

These results were consistent with their preferred visiting time, such as mornings for fog and evenings for sunsets. Most images' color was a dark greyish yellow, often evoking negative emotions, such as sadness, despair, fear, and humiliation, as shown in the study of the relationship between color and emotion. The characteristics of wetlands expressed in gray can also additionally deliver bad connotations. Nevertheless, based only on the colors of the photos, we cannot conclude that visitors' beliefs of the wetlands are bad. On the other hand, analysis of facial expressions of emotion, such as joy, surprise, sadness, anger, disgust, and fear of photos taken in the Oppo and Sancheon Bay lagoons, showed that most visitors smile in the pictures, expressing happiness.

Bouahim et al. [151] investigated the social perception of ecosystem services and vulnerability analysis associated with temporary-flooded wetlands. In this study, 101 ponds were randomly selected in Morocco. Information from each pond was gathered, including geographic coordinates, size, nature of the surrounding environment, ownership, land uses, proximity to roads/tracks, distance to habitats, and distance to the city. A vulnerability index was estimated for each pond and land use to evaluate, quantify, rate, and map threats in the study area.

This method was adopted by conducting interviews with well-known experts in ecology and social economics of natural habitats. Data related to the impact of human activities, such as grazing, recreation, drainage, cultivation, and urbanization on the plant communities were collected in 32 temporary ponds in the study area and used to define the indicators. For a socio-economic perception survey, 110 random households living near

ponds in three municipalities were selected. Interviews were performed from May to July 2008 using questionnaires that were previously tested in a pilot study.

The questionnaire consisted of five main sections: (i) socio-economic status of households, (ii) information about ponds and the services they provide, (iii) land use according to the local population, (iv) land ownership and guardianship, and (v) threat perception. Most interviewees stated that they observed the destruction of ponds over the past decade. The main threats perceived by the participants included the reduction of forage and water resources (threat to species), the use of ponds to drain and uncontrolled water pollution (threat of ecosystem destruction), and the extraction of building materials, urbanization and agriculture (threat to eliminate the ecosystem).

A total of 60% of respondents found ponds very useful and highlighted place attachment and cultural values. There were 19% of respondents who stated that ponds were not important and did not provide any services to the community. The threat map demonstrated that 22% of ponds were vulnerable and 23% were threatened in the short term.

Threat evaluation and mapping indicated higher threat levels for ponds located in farm landscapes and private land. This study showed that the disappearance of ponds after their conversion into agricultural land is part of eliminating the ecosystem. Ponds are considered areas that can be transformed into farmland to feed the growing local population. Conversely, forest environments have protected the ponds because their number has remained constant for the same time. Therefore, government services' management of forest environments has been essential in preserving ponds for the past 50 years.

Rojas et al. [147] evaluated the social perception of ecosystem services in a coastal wetlands after the earthquake. They executed a semi-structured survey to analyze the social perception of the perceived benefits of ecosystem services, which was divided into three stages: (i) design a survey, (ii) determine the population and size of the sample, and (iii) data analysis. They conducted 175 surveys in the study area. The data were investigated using IBM SPSS (Statistical Package for the Social Sciences). Perception scores obtained on the Likert scale were determined using Cronbach's alpha coefficient. This method generally applies to all ecosystem services.

The results indicated that the highest value was related to cultural services. In addition, the study area as an ecosystem can provide opportunities for services related to recreation, local identity, science, and environmental education. In this survey, the perceived values for hazard regulation services varied by gender. Accordingly, this issue is crucial for women. This difference in perception of gender could be related to the post-disaster economic effects that have significantly increased the unemployment rate among women.

In addition, the high frequency of natural disasters, such as tsunamis and floods, can positively affect the population's beneficial valuation of the wetland. Since wetlands are essential for mitigating the effects of such catastrophes, they are socially valued as post-earthquake recovery spaces, especially if wetlands provide biophysical properties and biodiversity, including water, bird species, micromammals, and amphibians.

Cottet et al. [141] investigated the relationship between human perception of wetland aesthetics and healthiness with wetlands ecological functioning in France. To this end, a photo questionnaire was provided to 403 laypeople and self-identified experts from Ain River wetlands. Two objectives were defined: (1) identification of various parameters, both visual and ecological, that affect the perception of the value of these ecosystems; (2) comparison of the perceptions of experts and laypeople. A questionnaire containing 16 photos of the wetland was presented to the people, and they were asked to evaluate each image according to the two perceptual criteria of aesthetics and environmental health. In addition, a set of photographs focusing on the water was collected from the study area. Photos were selected and classified based on two criteria:

(*i*) A visual criterion (4 classes): reflective water, transparent water with visible substrate, water with floating aquatic vegetation, and water with aquatic vegetation growing under the water's surface.

(*ii*) An ecological parameter about the nutritional status of the wetland (three classes): eutrophic, mesotrophic, and oligotrophic.

Participants were asked to use an "analogous visual scale" to rate the photos. This scale was first developed in medicine to treat pain and has recently been used to study landscape perception. The scale is continuous and limited to evaluation terms ("very high aesthetic value" vs. "no aesthetic value", "very healthy" vs. "not at all healthy"). Participants were asked to point out their evaluation on a relevant scale. Intergroup comparisons were used to analyze the effect of visual and ecological parameters on participants' perceptions, i.e., assessing the aesthetics and healthiness of the wetland.

The relationship between the environmental characteristics of wetlands and their understanding was analyzed with parametric statistical tests (ANOVA test and Tukey test). This method is a factorial analysis of correspondence about the occurrence of word classes used to describe wetlands. As a result, the study showed that people's perceiving of the aesthetics of the wetland strongly depends on visual criteria. The perceptions of laypeople and experts were equally affected by these parameters. The study revealed certain specific features of wetlands affecting perception. These included the transparency and color of the water, the presence and appearance of aquatic vegetation, and the presence of sediment in aquatic habitats.

Torres-Lima et al. [152] studied the socio-economic, environmental, and regional economic management variables that affect the population's perceptions of an urban wetland in Mexico City. As a methodology, this study used an approach that focused on area characteristics and fieldwork was conducted between April and May 2010. An initial pre-test was used to determine the main factors. In this test, factors or elements were considered that limit or define decisions or sustainability conditions from a combination of social processes and their interaction with the environment. These factors were included in the final survey as variables.

Sociodemographic data were collected using a survey modified from a pre-test on three factors that boost and restrict wetland sustainability. These factors included socioeconomic, environmental, and regional data and on perceived agendas for urban wetland planning. Locals were asked to express their level of perception of the sociological-economic implications for each term (where 1 = strongly, 2 = slightly, and 3 = little). This study showed that the functional characteristics of the wetland landscape were interpreted from different perspectives according to five major factors of practical statistical importance: age, sex, education, ownership, and indigenousness.

In peri-urban areas, agriculture is a potential source of employment and income without formal education requirements. The competence of farmers for urban or agricultural work is still associated with economic benefits for regional development. Therefore, education helps to create a positive local perception of urban wetlands. Education happens by enabling people to receive and interpret information about their environment and regional context.

For some participants, evaluation of the urban landscape was based on the view of aesthetic qualities and its potential use as a reserve for human habitation. On the other hand, the spatial organization of natural resources for environmental and agricultural production purposes makes it possible for other major populations to manage their ruralurban territory.

Manuel [153] investigated public awareness and perceptions of small urban wetlands. The research was a case study to examine the response to urban wetlands. Different sites were selected to identify similarities and differences in response to varying types of wetlands in different neighborhood settings. Three wetlands were chosen from 18 candidate sites that met the following criteria: less than 2.0 hectares, separate from adjacent water systems, natural and under "unmanaged" conditions, and located in an easily identifiable urban or peri-urban residential neighborhood.

In this study, 65 households were selected for each neighborhood, and interviews were conducted in summer and late spring 1996–97. The questionnaire consisted of 30 main

questions organized into four sections; a combination of yes/no questions, open-ended questions, and Likert-scaled (1 to 5) prescribed statements. The questionnaire asked participants about:

- (*i*) demographic information and household orientation to environmental issues and outdoor activities;
- (*ii*) knowledge or perception of the boundaries of the neighborhood, community character, and the natural areas in the neighborhood;
- specific information regarding the wetland site, including awareness, knowledge, and observation of the wetland; use of the wetland; understanding of the advantages or disadvantages related to wetlands; and attitudes toward or perceptions about wetlands; and
- *(iv)* general orientation to the values of "naturalness" and "wetlandness" in an urban environment.

The text of some questions allowed participants to incorporate the experience of other households in some of the answers. The study showed that participants were interested in the environment and recreation. Residents liked their neighborhoods, and they had a positive view of urban nature, especially their local wetlands. However, despite their positive estimation of environmental considerations and their agreement with the benefits of urban natural sites, these people paid little attention to the small wetlands.

On the other hand, they did not completely ignore or have unkind feelings to the wetlands, and they were not hostile to the wetlands. People who reported using the areas (almost half of the participants) explained using them for different purposes (skating, catching frogs, picking flowers, socializing, and enjoying being in a natural place). They widely accepted wetlands as part of their neighborhoods and enjoyed the areas.

Rooney et al. [55] compared natural wetlands with stormwater wetlands and stormwater ponds to examine resident perceptions of the environmental services of natural wetlands and stormwater ponds with their biophysical values. For the study, 72 wetlands were selected, belonging to four types: natural wetlands, natural wetlands impacted by agriculture, stormwater wetlands, and stormwater ponds. Each subset consisted of six wetlands, and three sample sets of 20–34 people were considered for an interview to survey social values.

The survey participants reported that they assessed many of the values of ecosystem services in the natural wetlands, agricultural wetlands, and stormwater management facilities they visited. More than 80% of the participants reported that they value biodiversity and groundwater recharge services for natural wetlands, which is, in fact, commensurate with the biophysical assessment. Remarkably, nearly 90% of the participants found value in the aesthetics of the stormwater pond, despite supporting few songbirds or other animals and having a simple strip of mainly invasive, upland vegetation. However, participants were largely mistaken (57%) in believing that groundwater recharge was valuable in impermeable stormwater ponds.

Participants gained almost as much value (slightly more than 68%) in natural wetlands (75%) as in stormwater ponds. Overall, more than 90% of study participants recognized the flood control provided by storm ponds, the main reason for their existence. Respondents in the natural heritage of natural wetlands (40%) found similar value to storm ponds (49%).

Pedersen et al. [154] identified whether peri-urban and urban wetland areas contributed to the well-being and quality of life of the surrounding residents and investigated their value compared to two other types of green spaces (i.e., parks and urban forests). The study evaluated the perception of local people of wetland areas inside or near their neighborhoods. This survey was conducted using a questionnaire. This research was based on proper environmental psychology tools, distributed by postal questionnaire in three municipalities with wetland areas of different structures and locations among residents. The data were analyzed statistically in IBM SPSS. A total of 473 people participated in this survey and completed the questionnaire.

It appeared that residents perceived wetland areas as contributing to their quality of life. Participants responded that the wetland area contributes to multiple aspects of quality

of life, such as dealing with nature and experiencing aesthetics. These areas also promote activities that support well-being. In addition, these areas appeared of high repair quality and evoked positive responses. Furthermore, recognizing and indicating the value of cultural ecosystem services can increase the total value attributed to urban wetland areas.

All wetland areas were rated high on most of the measured concepts. Still, their value relative to other green areas differed, possibly depending on the accessibility of the wetland and the availability of other green areas. The place and extent of integration of the wetland area into the residential area determine which aspects of quality of life are most satisfactory. Wetland areas can be attributed to the value of cultural ecosystem services based on how residents perceive their contribution to their quality of life. These values can be added to provisional and regulating ecosystem services and form the basis of planning for urban environments.

In summary, effective management, maintenance, and protection of wetlands are directly related to the positive public perceptions about urban wetlands values. Therefore, to improve public perceptions, there is a need for communication between people and urban governors, providing public awareness about urban wetlands and their services and values. Furthermore, enhancing public perceptions about the benefits of wetlands can support the protection and maintaining these natural and valuable resources.

8. Summary and Research Gaps

In this section, we summarize the reviewed topics in this article. We also address the research gaps and directions, where we identified during our review.

8.1. Urban Wetlands and Biodiversity

Land-use changes, from natural habitat to agricultural or urban land, are the primary factor that reduces biodiversity in urban wetlands. Most of the studies reviewed in this article examined the biodiversity of fauna and flora in wetlands at different time intervals in different years. These studies, for example, monitored vegetation establishment, water quality improvement, and animal colonization [69]. In addition to providing biodiversity, urban wetlands offer a wide range of ecosystem services: urban wetlands, e.g., they create a network of distributed and discrete habitat patches that facilitate the movement of species through the landscape [92,155].

Urban wetlands indeed contain a more significant proportion of biodiversity than other habitats. Therefore, ecological restoration allows contributing to the conservation of endangered species that have specific habitat requirements. The interaction between biodiversity and species is beneficial for human societies [86,90,91]. In addition, the interaction between humans and urban wetlands biodiversity stands as a natural and anthropological process. To improve people's interaction and wetland biodiversity, adaptive wetlands management approaches, policies and practices are needed [90].

Despite the valuable and massive work performed in the literature, the reviewed research in this article highlights the following research gaps concerning the linkage between biodiversity, humans, and the urban wetlands:

- (i) Wetlands also are considered as ecological traps. What are the challenges associated with these ecological traps? [82].
- (ii) How can we effectively manage constructed wetlands in urban landscapes? [82].
- (iii) How do constructed (e.g., treatment) wetlands play an essential role in contributing to biodiversity locally and globally? [87].

Answering these research gaps would require effective management and preservation of wetlands. However, to improve the biodiversity in wetlands, research should aim to obtain novel and proper management plans specifically for urban wetlands [87]. For example, the studies that focus on continuous restoration and maintenance programs can be promoted to ensure that urban wetlands function efficiently [82]. As a result, the efficient functions of urban wetlands would enrich biodiversity and solve ecological traps by improving the habitat quality for animals living in these wetlands.

On the other hand, wetlands as ecological traps result directly from inefficient management activities. Thus, prevention actions, such as reducing pollutant exposure, periodic urban wetland restoration services, and design management programs for high-risk species [87] can overcome the challenges associated with ecological traps and therefore improve the biodiversity in urban wetlands.

8.2. Urban Wetlands and Urban Heat Islands

The UHI effect directly relates to the current climate change, which has an influential impact on global warming. The studies surveyed in this article reported the reduction of temperature in places near blue spaces (such as wetlands, ponds, lakes, and rivers) in urban regions [65,101]. The temperature reduction is due to the cooling effects of these blue spaces, which have different features, such as scale, location, shape, geometry, distance, and hydrology [4,101].

Indeed, features, such as the numbers, sizes, and locations of blue spaces, specify the amounts and levels of the cooling effects in urban areas. For example, a large blue space was found to provide a powerful cooling effect on the surrounding area, while a few small lakes affected a higher percentage of the city [108].

Due to the crucial roles of blue spaces in reducing UHI, special attention should be given to these blue spaces within urban studies. Indeed, governments and urban planners can use the blue space resources in cities to plan and design sustainable living environments with minimizing UHIs in urban areas [13,65,108].

However, most of the studies in the literature aimed at finding solutions for reducing UHI in cities, yet there are a great number of challenges associated with UHI as follows:

- (i) How do blue spaces with different features (such as shapes and sizes) have considerable cooling effects on local thermal environments? What is the impact of an individual blue space, such as a wetland or a lake in a city? [107].
- (ii) A higher distance from the blue space means a greater average surface temperature and weaker wetland regulation. How do UHIs vary with the increase of distance from a wetland? [65].
- (iii) How do wetlands' shapes and positions impact the cooling effects of wetlands? [105].
- (iv) Urban wetlands have a more powerful cooling effect than green spaces. What is the reason that high-density buildings and riverbanks limit the spatial extent of the cooling effect? [101].
- (v) The type of land cover (such as vegetation and constructions) around blue spaces also significantly impacts UHI changes. How does the type of land cover around blue spaces impact UHIs in urban areas? [13].
- (vi) When green spaces, e.g., forests and blue spaces, are located beside each other, why do they create a mutually dependent environment and offer synergistic cooling? [4].
- (vii) Why and how is properly weighting the sizes and locations of blue spaces helpful for urban planning? [108].

Overcoming these challenges would require knowledge on wetlands characteristics, such as their shape, size, location, and distance from other urban spaces, such as forests, lakes, rivers, and built environments. As reviewed in this paper, some studies have taken these features into account when evaluating and discussing the cooling effects of wetlands.

For instance, while the land cover around wetlands directly impacts the cooling effect of wetlands [13], the compact shape of wetlands and square- and round-shaped wetlands can reduce the environmental temperature better than elongated or elongated irregular shapes [107]. In addition, the cooling effects of the wetlands have been realized to be significant close to cities [105]. However, in the literature, there are limited studies on the relation between urban wetlands and UHI characteristics, and therefore performing research on the topic is highly important. This importance stems from the current global warming situation.

8.3. Urban Wetlands and Urban Sustainability

Our article was focused on studies regarding the role of wetlands (whether natural or constructed) in increasing urban sustainability. Different types of water space, such as wetlands, ponds, river basins, and marshes, play an essential role in improving environmental sustainability. These promote the urban well-being of local environments and enable economic development [117,118].

The studies reviewed in this article highlight the importance and impact of water bodies, such as wetlands and ponds, rivers, basins, or marshes, providing sustainability and enabling the adaptation of urban areas to climate change. These water bodies affect urban microclimates and provide ecosystem services [115,116,119]. The studies emphasized the development and protection of wetlands in urban areas since wetlands are nature-based solutions that create livable and sustainable cities that are resilient in the face of climate change [58,116,117]. In addition, to increase urban sustainability, due to the resilience of wetland ecosystems, studies suggested that a balance should be established between land use and wetland rehabilitation in urban ecosystems [114].

Indeed, many studies are aiming at solving the challenges associated with urban sustainability. However, these studies address the following research gaps for further exploration:

- (i) What transformations are created when enclosed coastal wetlands are implemented in these environments? [114].
- (ii) How does afforestation contribute to the conservation of soil and water in urban wetlands? [58].
- (iii) How does ecotourism, such as bird watching, improve environmental conditions and benefit economic spillovers? [117].
- (iv) How does aquaculture implementation in environmentally sensitive areas reduce the extra constraints in financial and human resources? [117].
- (v) How does wetland vegetation, such as mangroves, help reduce tsunami damage in coastal wetlands in urban areas? [116].

Exploring these research gaps would require investigating the functionalities of wetlands in urban areas, which can help create liveable and sustainable cities. Among the diverse functionalities of wetlands, ecotourism is one of the best sources of income [114]. Therefore, ecotourism provided by wetlands, such as bird watching, can help the conservation of urban wetlands by spending the payments for maintaining and improving wetlands functionalities. In addition, developing local aquaculture, for instance, in coastal wetlands, would facilitate a long-term sustainable economic activity for locals by creating a local food economy.

By applying appropriate management strategies, the established economy can be used to benefit the wetlands and aquatic environments and provide sustainable environments [117]. Therefore, obtaining answers for the research questions on this topic by considering the functionalities of wetlands in urban settings would promote achieving a balance, for example, between economic activities and the protection of wetlands. This would offer efficient management strategies and hence provide sustainable urban environments.

8.4. Urban Wetlands and Recreational Values

The studies reviewed in this article analyzed wetlands as valuable opportunities for nature-based recreation. Studies emphasized the importance of raising public awareness about the conservation of water bodies, such as wetlands in the urban context. The reviewed articles highlighted the role and significance of water bodies, such as wetlands in providing opportunities for nature-based recreation services [31]. Some of the studies indicated that the locations of water bodies are essential for defining the recreational opportunities in urban areas [131]. The extensive recreational opportunities provided by wetlands included fishing, swimming, bird watching, and walking, which positively affect people's physical and mental health [129–131].

The recreational opportunities offered by water bodies also provide economic benefits in urban areas. Therefore, most of the studies used travel cost methods to estimate the monetary values of water bodies. Travel cost methods are based on the cost of travel to recreational sites and are widely used in the economic evaluation of the benefits of outdoor recreations [125,127–131,156]. Moreover, the studies state that the financial crisis does not directly impact the economic valuation of the recreational aspects of these wetlands [125]. However, reduced water quality decreases recreational activity, such as swimming, but does not affect activities, such as boating and fishing [129–131]. In addition, improving water quality to the extent that it supports wildlife and fisheries would increase economic benefits [127,132].

Indeed, creating a positive attitude toward wetlands and raising awareness about wetlands' social and cultural benefits in cities has an essential role in forming a strong desire to visit wetlands. Therefore, public awareness about the recreational benefits would support protecting and maintaining these natural and valuable resources [45,133].

In addition to the valuable works and results presented in the surveyed articles, these studies suggested the following as potential research gaps and future research topics:

- (i) Studies suggested peoples' incomes as an indicator to estimate the economic values of wetland. Is considering incomes an appropriate indicator? [125].
- (ii) What are the other approaches except for the travel cost method to analyze the economic value of wetland recreational services? [125].
- (iii) What are the missing links between governance partnerships and local people to develop ecotourism and conserve wetlands at the national and international levels? [132].
- (iv) How can considering some behavioral variables, such as motivation, satisfaction, experience, and place attachment, help better understand wetland visitors' attitudes and behaviors? [133].
- (v) How do different wetland environmental settings or geographical locations affect visitor behavior when visiting wetlands? [133].

Finding answers to these questions requires paying attention to the recreational values of wetlands as a nature-based solution. The recreational use of wetlands, e.g., ecotourism, is aligned with wetland conservation, as ecotourism promotes the local economy through local resources (i.e., wetlands). Therefore, managing the local economy can benefit the preservation of urban wetlands and significantly improve their recreational values [132]. In addition, behavioral variables, such as motivation, satisfaction, experiences, and perceptions about wetlands as well as travel cost methods can be applied to assess the recreational values of urban wetlands [125,133].

However, despite the findings in the existing studies, further research is needed to understand the parameters affecting the recreational values offered by urban wetlands as some studies argue about the inefficiency of applying specific approaches, such as travel cost methods. Moreover, since wetlands offer diverse recreations for people, wetlands can provide well-being and healthier living environments due to the emergence of mega-cities.

8.5. Urban Wetlands and Social Perceptions

The reviewed studies frequently used a questionnaire for evaluating people's perceptions about wetlands. However, the number of articles that used questionnaires to study water bodies is limited compared to the studies that assessed green spaces, such as forests and parks. To this end, there is a great opportunity for conducting questionnaire-based studies for water bodies.

The perception factors used in the studies provide a platform for natural resource managers and urban planners to improve urban wetland accessibility, visitor experience, and perception of safety for all users [157]. The reviewed articles reported that specific characteristics of wetlands, including water quality, transparency, and color of water, presence, the appearance of aquatic vegetation, and the presence of sediment in aquatic habitats, affect peoples' perceptions [55,141,150].

Visitors of urban forests and natural wetlands paid more attention to the opportunity to experience nature, their sense of the place, and place attachment. The visitors of constructed wetlands paid more attention to the quality or a particular activity [147]. Moreover, the studies also found that personal factors, such as peoples' gender and their experience with wetlands were more influential than environmental and social factors, such as the locations and recreational opportunities offered by wetlands. For example, females had more fear of spending their leisure time in places with crime(s) [149].

Due to the importance of peoples' social perception of wetlands in urban areas, there is a great need for further research. Hence, the studies in the literature suggest exploring the following research questions:

- How can a better understanding of urban wetlands' social values and perceptions help decision-makers achieve their social and environmental management goals? [149].
- (ii) What educational and interpretation programs are useful to increase the tendency for sustainable management of wetlands through wetland rehabilitation projects? [150].
- (iii) How can improving wetland conservation management by linking natural science (vulnerability assessment) with social sciences (understanding ecosystem services) help to improve the social perception of urban wetlands? [151].
- (iv) How does identifying the value of cultural ecosystem services that enable increasing urban wetlands' values help in understanding social perceptions? [154].

Exploring these research questions requires considering the social perceptions about urban wetlands. To improve people's perception of wetlands and increase public awareness about the importance of wetlands and their ecosystem services, there is the need for an efficient management strategy that involves individuals and all relevant stakeholders [149]. Furthermore, wetlands should be considered as important resources that offer tangible benefits for the economy, biodiversity, and local communities.

The management, policies, educational programs, and urban planning should integrate these natural resources into the urban environment [154]. Further research is needed to realize the causes affecting people's perception of urban wetlands and further illustrate the linkage between the perception and wetlands. More research on the topic would enable efficient management strategies and urban planning alongside the increase of urbanization, thus, supporting livable cities.

9. Discussion and Conclusions

Wetlands offer a wide range of ecosystem services. In urban areas, wetlands offer many benefits, such as cooling the urban environment, providing habitats for wildlife, recreational opportunities, water quality improvement, and mitigation for urban areas to cope with the effects of climate change. Wetlands located in urban and peri-urban areas, especially, play an important role in urban sustainability by providing valuable services. However, urban wetlands are usually not included in urban planning decisions, which often leads to weak governance. Urban wetlands need protection, restoration, and management to maintain the valuable services they provide.

Urban wetlands are more vulnerable to human intervention and destruction than any other ecosystem. The main drivers of the destruction and extinction of urban wetlands include population growth, uncontrolled urban construction, eutrophication, contamination, land conversion, drainage, changed water regime, over-exploitation, and biodiversity loss due to invasive alien species. It is estimated that climate change has led to the extinction and destruction of many wetlands, reducing their species and increasing the growing pressure on wetland ecosystems. Thus, growing pressures will reduce the functionality of wetlands to mitigate the effects of climate change and result in further reductions in human well-being.

In addition, the potential and help of stakeholders and individual citizens or groups of volunteers interested in cooperating in the development and maintenance of water environments bring innumerable ecological and social benefits to the urban environment. In the same way, with the help of academics, stakeholders, and local actors, governments conduct appropriate decisions and implement nature-based solutions to solve the intense environmental, economic, and social problems sourced from grand challenges, such as those posed by climate change.

In conclusion, urban wetlands should be managed sustainably by involving individuals and all relevant stakeholders as well as increasing public awareness about the importance of wetlands and their ecosystem services and benefits. In addition, policies, urban planning, and management should consider wetlands as blue infrastructures and integrate wetlands into the urban environment as they offer tangible benefits for the economy, biodiversity, and local communities. In short, policies should support and improve urban wetlands since they play an essential role in achieving sustainability in cities.

In this article, we reviewed the theoretical background of wetlands in urban areas, i.e., constructed wetlands and their relations to ecosystems values offered for urban environments and inhabitants. These values included the *sustainability*, *biodiversity*, *urban heat islands*, *social perception*, and *recreation* benefits offered to people in cities.

We systematically evaluated the role of wetlands in an urban environment. Specifically, for each study, we reviewed the objectives, methodologies, and findings. Moreover, we summarized the critical research gaps addressed in the reviewed articles, highlighted the significant open research challenges, and addressed future research directions in this field of study. For example, as potential research directions, we found less research on the consideration of wetlands in urban planning decisions and using this green-blue infrastructure as a solution to adapt to climate change, both of which are important for understanding the impact of urban wetlands.

As future works, we plan to explore improving the condition of wetlands by assessing resident perceptions of urban wetlands. To this end, we plan to collect questionnaire-based data from residents of a district in Helsinki regarding the functionalities of urban wetlands and propose approaches for planning and redesigning an old pond as a new urban wetland. We also plan to study the history of urban wetlands in Helsinki for almost the past century and evaluate their functions and services to learn lessons about the parameters sustaining these wetlands. In addition, we aim to identify the existing policies for maintaining wetlands in urban areas and propose methods and practices for improving these policies and decision-making.

Author Contributions: S.A. was involved in conceptualization, planning and writing the manuscript, designing the layout and the structure, editing and revising the manuscript. P.N. helped in conceptualization, supervised, edited and revised the manuscript. A.O. co-supervised, reviewed and revised the manuscript. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Acknowledgments: The authors would like to thank the library of the University of Helsinki (HULib) for supporting the article processing charge of this article.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. United Nations. *World Urbanization Prospects: The 2018 Revision;* Technical Report; United Nations, 2018. Available online: https://www.un-ilibrary.org/content/books/9789210043144 (accessed on 8 November 2021).
- 2. Ampatzidis, P.; Kershaw, T. A review of the impact of blue space on the urban microclimate. *Sci. Total Environ.* **2020**, 730, 139068. [CrossRef]
- 3. Liao, R.; Jin, Z.; Chen, M.; Li, S. An integrated approach for enhancing the overall performance of constructed wetlands in urban areas. *Water Res.* 2020, *187*, 116443. [CrossRef] [PubMed]
- 4. Gunawardena, K.R.; Wells, M.J.; Kershaw, T. Utilising green and bluespace to mitigate urban heat island intensity. *Sci. Total Environ.* **2017**, *584*, 1040–1055. [CrossRef]
- 5. Kati, V.; Jari, N. Bottom-up thinking—Identifying socio-cultural values of ecosystem services in local blue–green infrastructure planning in Helsinki, Finland. *Land Use Policy* **2016**, *50*, *537–547*. [CrossRef]
- 6. Ramsar, P. The Ramsar Convention Manual: A guide to the Convention on Wetlands (Ramsar, Iran, 1971) Ramsar Convention Secretariat. 2013. Available online: https://aquadocs.org/handle/1834/330 (accessed on 8 November 2021).
- 7. Boyer, T.; Polasky, S. Valuing urban wetlands: A review of non-market valuation studies. Wetlands 2004, 24, 744–755. [CrossRef]

- 8. Temmerman, S.; Meire, P.; Bouma, T.J.; Herman, P.M.; Ysebaert, T.; De Vriend, H.J. Ecosystem-based coastal defence in the face of global change. *Nature* **2013**, *504*, 79–83. [CrossRef]
- 9. Quin, A.; Jaramillo, F.; Destouni, G. Dissecting the ecosystem service of large-scale pollutant retention: The role of wetlands and other landscape features. *Ambio* 2015, 44, 127–137. [CrossRef] [PubMed]
- 10. Verhoeven, J.T.; Arheimer, B.; Yin, C.; Hefting, M.M. Regional and global concerns over wetlands and water quality. *Trends Ecol. Evol.* **2006**, *21*, 96–103. [CrossRef] [PubMed]
- 11. Mitsch, W.J.; Bernal, B.; Nahlik, A.M.; Mander, Ü.; Zhang, L.; Anderson, C.J.; Jørgensen, S.E.; Brix, H. Wetlands, carbon, and climate change. *Landsc. Ecol.* 2013, 28, 583–597. [CrossRef]
- 12. Das, A.; Basu, T. Assessment of peri-urban wetland ecological degradation through importance-performance analysis (IPA): A study on Chatra Wetland, India. *Ecol. Indic.* 2020, 114, 106274. [CrossRef]
- 13. Şimşek, Ç.K.; Ödül, H. Investigation of the effects of wetlands on micro-climate. Appl. Geogr. 2018, 97, 48–60. [CrossRef]
- 14. Seifollahi-Aghmiuni, S.; Nockrach, M.; Kalantari, Z. The potential of wetlands in achieving the sustainable development goals of the 2030 Agenda. *Water* **2019**, *11*, 609. [CrossRef]
- Bennett, G.; Mulongoy, K.J. Review of Experience with Ecological Networks, Corridors and Buffer Zones; Technical Series; Secretariat of the Convention on Biological Diversity: Montreal, QC, Canada, 2006; Volume 23, p. 100.
- 16. Zhou, J.; Wu, J.; Gong, Y. Valuing wetland ecosystem services based on benefit transfer: A meta-analysis of China wetland studies. *J. Clean. Prod.* **2020**, 276, 122988. [CrossRef]
- 17. Bassi, N.; Kumar, M.D.; Sharma, A.; Pardha-Saradhi, P. Status of wetlands in India: A review of extent, ecosystem benefits, threats and management strategies. *J. Hydrol. Reg. Stud.* **2014**, *2*, 1–19. [CrossRef]
- 18. Finlayson, C.; Davidson, N. Global review of wetland resources and priorities for wetland inventory. *Pref. Summ. Rep.* **1999**, 15. Available online: https://www.ramsar.org/node/23484 (accessed on 8 November 2021).
- 19. Casagrande, D.G. The human component of urban wetland restoration. *Restor. Urban Salt Marsh: Interdiscip. Approach* **1997**, *100*, 136–150.
- Kometa, S.; Kimengsi, J.; Petiangma, D. Urban development and its implications on wetland ecosystem services in Ndop. *Environ.* Manag. Sustain. Dev. 2017, 7, 12141. [CrossRef]
- Thorslund, J.; Jarsjo, J.; Jaramillo, F.; Jawitz, J.W.; Manzoni, S.; Basu, N.B.; Chalov, S.R.; Cohen, M.J.; Creed, I.F.; Goldenberg, R.; et al. Wetlands as large-scale nature-based solutions: Status and challenges for research, engineering and management. *Ecol. Eng.* 2017, 108, 489–497. [CrossRef]
- 22. Turner, R.; Van den Bergh, J.; Barendregt, A.; Maltby, E. Ecological-Economic Analysis of Wetlands: Science and Social Science Integration. TI Discussion Paper. 1998. Available online: https://www.researchgate.net/publication/251611755_Ecological-economic_Analysis_of_Wetlands_Science_and_Social_Science_Integration (accessed on 8 November 2021).
- Stoll-Kleemann, S. The rationale of socio-economic research for the successful protection and use of wetlands: The example of participatory management approaches. *Hydrobiologia* 2004, 527, 15–17. [CrossRef]
- 24. Gardner, R.C.; Finlayson, M. Global Wetland Outlook: State of the World'S Wetlands and Their Services to People 2018; Secretariat of the Ramsar Convention: Gland, Switzerland, 2018.
- Shine, C.; De Klemm, C. Wetlands, Water, and the Law: Using Law to Advance Wetland Conservation and Wise Use; Number 38; IUCN, 1999. Available online: https://www.iucn.org/content/wetlands-water-and-law-using-law-advance-wetland-conservation-and-wise-use (accessed on 8 November 2021).
- 26. Grobicki, A.; Chalmers, C.; Jennings, E.; Jones, T.; Peck, D. An Introduction to the Ramsar Convention on Wetlands (Previously the Ramsar Convention Manual); Ramsar Convention Secretariat: Gland, Switzerland, 2016.
- Ye, P.; Hao, X.; Cao, Y. Analysis on Ecological Protection of Urban Wetland. *Nat. Resour. Conserv. Res.* 2018, 1. Available online: https://www.researchgate.net/publication/325725806_Analysis_on_Ecological_Protection_of_Urban_Wetland (accessed on 8 November 2021). [CrossRef]
- Bertule, M.; Appelquist, L.R.; Spensley, J.; Trærup, S.L.M.; Naswa, P. Climate Change Adaptation Technologies for Water: A Practitioner's Guide to Adaptation Technologies for Increased Water Sector Resilience. 2018. Available online: https://www.ctcn.org/resources/climate-change-adaptation-technologies-water-practitioner-s-guide-adaptation-technologies (accessed on 8 November 2021).
- 29. McInnes, R.J.; Everard, M. Rapid assessment of wetland ecosystem services (RAWES): An example from Colombo, Sri Lanka. *Ecosyst. Serv.* 2017, 25, 89–105. [CrossRef]
- 30. Yun, H.; Kang, D.; Kang, Y. Outdoor recreation planning and management considering FROS and Carrying capacities: A case study of forest wetland in Yeongam-gum, South Korea. *Environ. Dev. Sustain.* **2021**, 1–25. Available online: https://www.researchgate.net/publication/351061245_Outdoor_recreation_planning_and_management_considering_FROS_ and_Carrying_capacities_A_case_study_of_forest_wetland_in_Yeongam-gum_South_Korea (accessed on 8 November 2021).
- 31. Alikhani, S.; Nummi, P.; Ojala, A. Urban water bodies and recreational opportunities in Finland. In Proceedings of the 17th International Conference on Environmental Science and Technology (CEST2021), Athens, Greece, 1–4 September 2021; pp. 1–4.
- 32. Shahjahan, A.T.M.; Ahmed, K.S.; Said, I.B. Study on Riparian Shading Envelope for Wetlands to Create Desirable Urban Bioclimates. *Atmosphere* **2020**, *11*, 1348. [CrossRef]
- Alikhani, S. The Concept and Practice of Green-Blue Infrastructure in New Urban Area of Malmi Airport. 2018. Available online: https://www.theseus.fi/handle/10024/159294 (accessed on 8 November 2021).

- 34. Sleegers, F. Phytoremediation as green infrastructure and a landscape of experiences. In Proceedings of the Annual International Conference on Soils, Sediments, Water and Energy, Amherst, MA, USA, 18–21 October 2010; Volume 15, p. 13.
- 35. Ingrao, C.; Failla, S.; Arcidiacono, C. A comprehensive review of environmental and operational issues of constructed wetland systems. *Curr. Opin. Environ. Sci. Health* **2020**, *13*, 35–45. [CrossRef]
- 36. Well, F.; Ludwig, F. Blue–green architecture: A case study analysis considering the synergetic effects of water and vegetation. *Front. Archit. Res.* **2020**, *9*, 191–202. [CrossRef]
- 37. Perini, K.; Sabbion, P. Green and Blue Infrastructure in Cities. Urban Sustain. River Restor. 2017, 1–9.
- 38. Lin, W.; Yu, T.; Chang, X.; Wu, W.; Zhang, Y. Calculating cooling extents of green parks using remote sensing: Method and test. *Landsc. Urban Plan.* **2015**, *134*, 66–75. [CrossRef]
- 39. Yang, G.; Yu, Z.; Jørgensen, G.; Vejre, H. How can urban blue-green space be planned for climate adaption in high-latitude cities? A seasonal perspective. *Sustain. Cities Soc.* **2020**, *53*, 101932. [CrossRef]
- 40. Chan, S.; Chau, C.K.; Leung, T. On the study of thermal comfort and perceptions of environmental features in urban parks: A structural equation modeling approach. *Build. Environ.* **2017**, *122*, 171–183. [CrossRef]
- 41. Labib, S.; Lindley, S.; Huck, J.J. Spatial dimensions of the influence of urban green-blue spaces on human health: A systematic review. *Environ. Res.* **2020**, *180*, 108869. [CrossRef]
- Grellier, J.; White, M.P.; Albin, M.; Bell, S.; Elliott, L.R.; Gascón, M.; Gualdi, S.; Mancini, L.; Nieuwenhuijsen, M.J.; Sarigiannis, D.A.; et al. BlueHealth: A study programme protocol for mapping and quantifying the potential benefits to public health and well-being from Europe's blue spaces. *BMJ Open* 2017, *7*, e016188. [CrossRef]
- Venkataramanan, V.; Packman, A.I.; Peters, D.R.; Lopez, D.; McCuskey, D.J.; McDonald, R.I.; Miller, W.M.; Young, S.L. A systematic review of the human health and social well-being outcomes of green infrastructure for stormwater and flood management. J. Environ. Manag. 2019, 246, 868–880. [CrossRef] [PubMed]
- 44. McInnes, R.J. Recognising wetland ecosystem services within urban case studies. Mar. Freshw. Res. 2014, 65, 575–588. [CrossRef]
- 45. Odgaard, M.V.; Turner, K.G.; Bøcher, P.K.; Svenning, J.C.; Dalgaard, T. A multi-criteria, ecosystem-service value method used to assess catchment suitability for potential wetland reconstruction in Denmark. *Ecol. Indic.* **2017**, 77, 151–165. [CrossRef]
- 46. Song, F.; Su, F.; Zhu, D.; Li, L.; Li, H.; Sun, D. Evaluation and driving factors of sustainable development of the wetland ecosystem in Northeast China: An emergy approach. *J. Clean. Prod.* **2020**, *248*, 119236. [CrossRef]
- 47. Russi, D.; ten Brink, P.; Farmer, A.; Badura, T.; Coates, D.; Förster, J.; Kumar, R.; Davidson, N. *The Economics of Ecosystems and Biodiversity for Water and Wetlands*; IEEP: London, UK; Brussels, Belgium, 2013; Volume 78.
- Nesshöver, C.; Assmuth, T.; Irvine, K.N.; Rusch, G.M.; Waylen, K.A.; Delbaere, B.; Haase, D.; Jones-Walters, L.; Keune, H.; Kovacs, E.; et al. The science, policy and practice of nature-based solutions: An interdisciplinary perspective. *Sci. Total Environ.* 2017, 579, 1215–1227. [CrossRef] [PubMed]
- 49. Patra, S.; Sahoo, S.; Mishra, P.; Mahapatra, S.C. Impacts of urbanization on land use/cover changes and its probable implications on local climate and groundwater level. *J. Urban Manag.* **2018**, *7*, 70–84. [CrossRef]
- Davidson, N.C. How much wetland has the world lost? Long-term and recent trends in global wetland area. *Mar. Freshw. Res.* 2014, 65, 934–941. [CrossRef]
- 51. U.S. Environmental Protection Agency Office. *Natural Wetlands and Urban Stormwater: Potential Impacts and Management;* O. a. WWD Office of Wetlands, Ed.; U.S. Environmental Protection Agency Office: Washington, DC, USA, 1993.
- 52. Valk, A.v.d. Northern Prairie Wetlands; Iowa State University: Ames, IA, USA, 1989.
- 53. Kantrud, H.A.; Millar, J.B.; van der Valk, A.G. Vegetation of wetlands of the prairie pothole region. *North. Prairie Wetl.* **1989**, 132–187.
- 54. Sharma, R.; Vymazal, J.; Malaviya, P. Application of floating treatment wetlands for stormwater runoff: A critical review of the recent developments with emphasis on heavy metals and nutrient removal. *Sci. Total Environ.* **2021**, 777, 146044. [CrossRef]
- 55. Rooney, R.; Foote, L.; Krogman, N.; Pattison, J.; Wilson, M.; Bayley, S. Replacing natural wetlands with stormwater management facilities: Biophysical and perceived social values. *Water Res.* 2015, *73*, 17–28. [CrossRef]
- Guide, A.P.; Resilience, S. Climate Change Adaptation Technologies for Water. UN Climate Technology Centre and Network, Report. Available online: http://sdg.iisd.org/news/publications-provide-tap-guidance-highlight-adaptation-technologiesfor-water/ (accessed on 8 November 2021).
- Stefanakis, A.; Akratos, C.S.; Tsihrintzis, V.A. Vertical Flow Constructed Wetlands: Eco-Engineering Systems for Wastewater and Sludge Treatment; Newnes, 2014. Available online: https://www.sciencedirect.com/book/9780124046122/vertical-flow-constructedwetlands (accessed on 8 November 2021).
- Chen, V.Y.C.; Lin, J.C.L.; Tzeng, G.H. Assessment and improvement of wetlands environmental protection plans for achieving sustainable development. *Environ. Res.* 2019, 169, 280–296. [CrossRef]
- 59. Chen, R.Z.; Wong, M.H. Integrated wetlands for food production. Environ. Res. 2016, 148, 429–442. [CrossRef]
- Hammer, D.; Bastian, R. Wetlands ecosystems: Natural water purifiers: Constructed wetlands for wastewater treatment: Municipal, industrial and agricultural, v. 5. *Water* 1989, 519. Available online: https://www.routledge.com/Constructed-Wetlands-for-Wastewater-Treatment-Municipal-Industrial-and/Hammer/p/book/9780367450922 (accessed on 8 November 2021).
- 61. Zhang, H.; Cui, B.; Hong, J.; Zhang, K. Synergism of natural and constructed wetlands in Beijing, China. *Ecol. Eng.* **2011**, 37, 128–138. [CrossRef]

- 62. Zhang, W.; Jiang, J.; Zhu, Y. Change in urban wetlands and their cold island effects in response to rapid urbanization. *Chin. Geogr. Sci.* 2015, 25, 462–471. [CrossRef]
- 63. Masi, F.; Rizzo, A.; Regelsberger, M. The role of constructed wetlands in a new circular economy, resource oriented, and ecosystem services paradigm. *J. Environ. Manag.* 2018, 216, 275–284. [CrossRef]
- 64. Saeed, T.; Muntaha, S.; Rashid, M.; Sun, G.; Hasnat, A. Industrial wastewater treatment in constructed wetlands packed with construction materials and agricultural by-products. *J. Clean. Prod.* **2018**, *189*, 442–453. [CrossRef]
- 65. Chun-ye, W.; Wei-ping, Z. Analysis of the impact of urban wetland on urban temperature based on remote sensing technology. *Procedia Environ. Sci.* **2011**, *10*, 1546–1552.
- 66. Saeed, T.; Khan, T. Constructed wetlands for industrial wastewater treatment: Alternative media, input biodegradation ratio and unstable loading. *J. Environ. Chem. Eng.* **2019**, *7*, 103042. [CrossRef]
- 67. Varjani, S.; Pandey, A.; Tyagi, R.; Ngo, H.H.; Larroche, C. *Current Developments in Biotechnology and Bioengineering: Emerging Organic Micro-Pollutants;* Elsevier: Amsterdam, The Netherlands, 2020.
- Ghermandi, A.; Van Den Bergh, J.C.; Brander, L.M.; De Groot, H.L.; Nunes, P.A. Values of natural and human-made wetlands: A meta-analysis. *Water Resour. Res.* 2010, 46. Available online: https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2010 WR009071 (accessed on 8 November 2021). [CrossRef]
- Wahlroos, O.; Valkama, P.; Mäkinen, E.; Ojala, A.; Vasander, H.; Väänänen, V.M.; Halonen, A.; Lindén, L.; Nummi, P.; Ahponen, H.; et al. Urban wetland parks in Finland: Improving water quality and creating endangered habitats. *Int. J. Biodivers. Sci. Ecosyst. Serv. Manag.* 2015, 11, 46–60. [CrossRef]
- 70. Mander, U.; Jenssen, P. *Natural Wetlands for Wastewater Treatment in Cold Climates*; WIT, 2002. Available online: https://www. witpress.com/books/978-1-85312-859-2 (accessed on 8 November 2021).
- 71. Stefanakis, A.I. The role of constructed wetlands as green infrastructure for sustainable urban water management. *Sustainability* **2019**, *11*, 6981. [CrossRef]
- Mulkeen, C.J.; Gibson-Brabazon, S.; Carlin, C.; Williams, C.; Healy, M.G.; Mackey, P.; Gormally, M.J. Habitat suitability assessment of constructed wetlands for the smooth newt (Lissotriton vulgaris [Linnaeus, 1758]): A comparison with natural wetlands. *Ecol. Eng.* 2017, 106, 532–540. [CrossRef]
- 73. Darrah, S.E.; Shennan-Farpón, Y.; Loh, J.; Davidson, N.C.; Finlayson, C.M.; Gardner, R.C.; Walpole, M.J. Improvements to the Wetland Extent Trends (WET) index as a tool for monitoring natural and human-made wetlands. *Ecol. Indic.* 2019, *99*, 294–298. [CrossRef]
- 74. Mitsch, W.J.; Bernal, B.; Hernandez, M.E. Ecosystem services of wetlands. *Int. J. Biodivers. Sci. Ecosyst. Serv. Manag.* 2015, 11, 1–4. Available online: https://www.tandfonline.com/doi/full/10.1080/21513732.2015.1006250 (accessed on 8 November 2021).
- 75. Mao, D.; Luo, L.; Wang, Z.; Wilson, M.C.; Zeng, Y.; Wu, B.; Wu, J. Conversions between natural wetlands and farmland in China: A multiscale geospatial analysis. *Sci. Total Environ.* **2018**, *634*, 550–560. [CrossRef] [PubMed]
- 76. Mallick, P.H.; Chakraborty, S.K. Forest, wetland and biodiversity: Revealing multi-faceted ecological services from ecorestoration of a degraded tropical landscape. *Ecohydrol. Hydrobiol.* **2018**, *18*, 278–296. [CrossRef]
- 77. Eppink, F.V.; van den Bergh, J.C.; Rietveld, P. Modelling biodiversity and land use: Urban growth, agriculture and nature in a wetland area. *Ecol. Econ.* 2004, *51*, 201–216. [CrossRef]
- 78. Verones, F.; Saner, D.; Pfister, S.; Baisero, D.; Rondinini, C.; Hellweg, S. Effects of consumptive water use on biodiversity in wetlands of international importance. *Environ. Sci. Technol.* **2013**, *47*, 12248–12257. [CrossRef]
- 79. Herath, I.; Vithanage, M. Phytoremediation in constructed wetlands. In *Phytoremediation*; Springer: Berlin/Heidelberg, Germany, 2015; pp. 243–263.
- 80. Hassall, C. The ecology and biodiversity of urban ponds. Wiley Interdiscip. Rev. Water 2014, 1, 187–206. [CrossRef]
- 81. Blicharska, M.; Andersson, J.; Bergsten, J.; Bjelke, U.; Hilding-Rydevik, T.; Johansson, F. Effects of management intensity, function and vegetation on the biodiversity in urban ponds. *Urban For. Urban Green.* **2016**, *20*, 103–112. [CrossRef]
- Hale, R.; Swearer, S.E.; Sievers, M.; Coleman, R. Balancing biodiversity outcomes and pollution management in urban stormwater treatment wetlands. *J. Environ. Manag.* 2019, 233, 302–307. [CrossRef]
- 83. Mayer-Pinto, M.; Johnston, E.L.; Bugnot, A.B.; Glasby, T.M.; Airoldi, L.; Mitchell, A.; Dafforn, K.A. Building 'blue': An eco-engineering framework for foreshore developments. *J. Environ. Manag.* **2017**, *189*, 109–114. [CrossRef] [PubMed]
- 84. Soanes, K.; Sievers, M.; Chee, Y.E.; Williams, N.S.; Bhardwaj, M.; Marshall, A.J.; Parris, K.M. Correcting common misconceptions to inspire conservation action in urban environments. *Conserv. Biol.* **2019**, *33*, 300–306. [CrossRef]
- 85. Goertzen, D.; Suhling, F. Promoting dragonfly diversity in cities: Major determinants and implications for urban pond design. *J. Insect Conserv.* **2013**, *17*, 399–409. [CrossRef]
- Van Roon, M.R. Wetlands in The Netherlands and New Zealand: Optimising biodiversity and carbon sequestration during urbanisation. J. Environ. Manag. 2012, 101, 143–150. [CrossRef]
- 87. Semeraro, T.; Giannuzzi, C.; Beccarisi, L.; Aretano, R.; De Marco, A.; Pasimeni, M.R.; Zurlini, G.; Petrosillo, I. A constructed treatment wetland as an opportunity to enhance biodiversity and ecosystem services. *Ecol. Eng.* **2015**, *82*, 517–526. [CrossRef]
- 88. Liao, W.; Venn, S.; Niemelä, J. Environmental determinants of diving beetle assemblages (Coleoptera: Dytiscidae) in an urban landscape. *Biodivers. Conserv.* 2020, 29, 2343–2359. [CrossRef]
- 89. Gascón, S.; Boix, D.; Sala, J. Are different biodiversity metrics related to the same factors? A case study from Mediterranean wetlands. *Biol. Conserv.* 2009, 142, 2602–2612. [CrossRef]

- 90. Da Silva, C.J.; Sousa, K.N.S.; Ikeda-Castrillon, S.K.; Lopes, C.R.A.S.; da Silva Nunes, J.R.; Carniello, M.A.; Mariotti, P.R.; Lazaro, W.L.; Morini, A.; Zago, B.W.; et al. Biodiversity and its drivers and pressures of change in the wetlands of the Upper Paraguay–Guaporé Ecotone, Mato Grosso (Brazil). *Land Use Policy* 2015, 47, 163–178. [CrossRef]
- 91. Meli, P.; Benayas, J.M.R.; Balvanera, P.; Ramos, M.M. Restoration enhances wetland biodiversity and ecosystem service supply, but results are context-dependent: A meta-analysis. *PLoS ONE* **2014**, *9*, e93507. [CrossRef] [PubMed]
- 92. Morganti, M.; Manica, M.; Bogliani, G.; Gustin, M.; Luoni, F.; Trotti, P.; Perin, V.; Brambilla, M. Multi-species habitat models highlight the key importance of flooded reedbeds for inland wetland birds: Implications for management and conservation. *Avian Res.* **2019**, *10*, 1–13. [CrossRef]
- Guareschi, S.; Abellán, P.; Laini, A.; Green, A.J.; Sánchez-Zapata, J.A.; Velasco, J.; Millán, A. Cross-taxon congruence in wetlands: Assessing the value of waterbirds as surrogates of macroinvertebrate biodiversity in Mediterranean Ramsar sites. *Ecol. Indic.* 2015, 49, 204–215. [CrossRef]
- 94. Shirani-Bidabadi, N.; Nasrabadi, T.; Faryadi, S.; Larijani, A.; Roodposhti, M.S. Evaluating the spatial distribution and the intensity of urban heat island using remote sensing, case study of Isfahan city in Iran. *Sustain. Cities Soc.* **2019**, *45*, 686–692. [CrossRef]
- 95. Guhathakurta, S.; Gober, P. The impact of the Phoenix urban heat island on residential water use. *J. Am. Plan. Assoc.* 2007, 73, 317–329. [CrossRef]
- Wu, S.; Yang, H.; Luo, P.; Luo, C.; Li, H.; Liu, M.; Ruan, Y.; Zhang, S.; Xiang, P.; Jia, H.; et al. The effects of the cooling efficiency of urban wetlands in an inland megacity: A case study of Chengdu, Southwest China. *Build. Environ.* 2021, 204, 108128. [CrossRef]
- 97. Deilami, K.; Kamruzzaman, M.; Liu, Y. Urban heat island effect: A systematic review of spatio-temporal factors, data, methods, and mitigation measures. *Int. J. Appl. Earth Obs. Geoinf.* **2018**, *67*, 30–42. [CrossRef]
- 98. Kondo, K.; Mabon, L.; Bi, Y.; Chen, Y.; Hayabuchi, Y. Balancing conflicting mitigation and adaptation behaviours of urban residents under climate change and the urban heat island effect. *Sustain. Cities Soc.* **2021**, *65*, 102585. [CrossRef]
- 99. Nwakaire, C.M.; Onn, C.C.; Yap, S.P.; Yuen, C.W.; Onodagu, P.D. Urban Heat Island Studies with Emphasis on Urban Pavements: A review. *Sustain. Cities Soc.* 2020, *63*, 102476. [CrossRef]
- Li, X.; Li, W.; Middel, A.; Harlan, S.L.; Brazel, A.J.; Turner Ii, B. Remote sensing of the surface urban heat island and land architecture in Phoenix, Arizona: Combined effects of land composition and configuration and cadastral–demographic–economic factors. *Remote Sens. Environ.* 2016, 174, 233–243. [CrossRef]
- 101. Xue, Z.; Hou, G.; Zhang, Z.; Lyu, X.; Jiang, M.; Zou, Y.; Shen, X.; Wang, J.; Liu, X. Quantifying the cooling-effects of urban and peri-urban wetlands using remote sensing data: Case study of cities of Northeast China. *Landsc. Urban Plan.* 2019, 182, 92–100. [CrossRef]
- 102. Elliot, T.; Almenar, J.B.; Rugani, B. Modelling the relationships between urban land cover change and local climate regulation to estimate urban heat island effect. *Urban For. Urban Green.* **2020**, *50*, 126650. [CrossRef]
- 103. Santamouris, M. Recent progress on urban overheating and heat island research. Integrated assessment of the energy, environmental, vulnerability and health impact. Synergies with the global climate change. *Energy Build.* 2020, 207, 109482. [CrossRef]
- 104. Wu, D.; Wang, Y.; Fan, C.; Xia, B. Thermal environment effects and interactions of reservoirs and forests as urban blue-green infrastructures. *Ecol. Indic.* 2018, 91, 657–663. [CrossRef]
- 105. Sun, R.; Chen, A.; Chen, L.; Lü, Y. Cooling effects of wetlands in an urban region: The case of Beijing. *Ecol. Indic.* **2012**, *20*, 57–64. [CrossRef]
- 106. O'Malley, C.; Piroozfarb, P.A.; Farr, E.R.; Gates, J. An investigation into minimizing urban heat island (UHI) effects: A UK perspective. *Energy Procedia* 2014, *62*, 72–80. [CrossRef]
- 107. Lin, Y.; Wang, Z.; Jim, C.Y.; Li, J.; Deng, J.; Liu, J. Water as an urban heat sink: Blue infrastructure alleviates urban heat island effect in mega-city agglomeration. *J. Clean. Prod.* 2020, 262, 121411. [CrossRef]
- 108. Theeuwes, N.E.; Solcerova, A.; Steeneveld, G.J. Modeling the influence of open water surfaces on the summertime temperature and thermal comfort in the city. *J. Geophys. Res. Atmos.* **2013**, *118*, 8881–8896. [CrossRef]
- 109. Hathway, E.; Sharples, S. The interaction of rivers and urban form in mitigating the Urban Heat Island effect: A UK case study. *Build. Environ.* **2012**, *58*, 14–22. [CrossRef]
- 110. Steeneveld, G.J.; Koopmans, S.; Heusinkveld, B.G.; Theeuwes, N.E. Refreshing the role of open water surfaces on mitigating the maximum urban heat island effect. *Landsc. Urban Plan.* **2014**, 121, 92–96. [CrossRef]
- 111. Roy-Basu, A.; Bharat, G.K.; Chakraborty, P.; Sarkar, S. Adaptive co-management model for the East Kolkata wetlands: A sustainable solution to manage the rapid ecological transformation of a peri-urban landscape. *Sci. Total Environ.* 2020, 698, 134203. [CrossRef]
- 112. Su, F.; Liu, H.; Zhu, D.; Li, L.; Wang, T. Sustainability assessment of the Liaohe Estuary wetland based on emergy analysis. *Ecol. Indic.* **2020**, *119*, 106837. [CrossRef]
- Byeon, C.W.; Nam, B.E. An assessment of the ecological functions of a sustainable structured wetland biotope (SSB). *Ecol. Eng.* 2020, 145, 105723. [CrossRef]
- 114. Wu, W.; Yang, Z.; Tian, B.; Huang, Y.; Zhou, Y.; Zhang, T. Impacts of coastal reclamation on wetlands: Loss, resilience, and sustainable management. *Estuar. Coast. Shelf Sci.* 2018, 210, 153–161. [CrossRef]
- 115. Zhou, L.; Guan, D.; Huang, X.; Yuan, X.; Zhang, M. Evaluation of the cultural ecosystem services of wetland park. *Ecol. Indic.* **2020**, *114*, 106286.

- 116. Rojas, C.; Munizaga, J.; Rojas, O.; Martínez, C.; Pino, J. Urban development versus wetland loss in a coastal Latin American city: Lessons for sustainable land use planning. *Land Use Policy* **2019**, *80*, 47–56. [CrossRef]
- 117. Sousa, C.A.; Cunha, M.E.; Ribeiro, L. Tracking 130 years of coastal wetland reclamation in Ria Formosa, Portugal: Opportunities for conservation and aquaculture. *Land Use Policy* **2020**, *94*, 104544. [CrossRef]
- 118. Luo, S.z.; Zhang, H.y.; Wang, J.q.; Li, L. Group decision-making approach for evaluating the sustainability of constructed wetlands with probabilistic linguistic preference relations. *J. Oper. Res. Soc.* **2019**, *70*, 2039–2055. [CrossRef]
- 119. Varin, M.; Theau, J.; Fournier, R.A. Mapping ecosystem services provided by wetlands at multiple spatiotemporal scales: A case study in Quebec, Canada. *J. Environ. Manag.* **2019**, 246, 334–344. [CrossRef]
- 120. Moeltner, K.; Woodward, R. Meta-functional benefit transfer for wetland valuation: Making the most of small samples. *Environ. Resour. Econ.* **2009**, *42*, 89–108. [CrossRef]
- 121. Gürlük, S.; Rehber, E. A travel cost study to estimate recreational value for a bird refuge at Lake Manyas, Turkey. *J. Environ. Manag.* **2008**, *88*, 1350–1360. [CrossRef]
- 122. Ward, F.A.; Beal, D. *Valuing Nature with Travel Cost Models*; Edward Elgar Publishing, 2000. Available online: https://www.e-elgar.com/shop/gbp/valuing-nature-with-travel-cost-models-9781840640786.html (accessed on 8 November 2021).
- 123. Ye, S.; Laws, E.A.; Costanza, R.; Brix, H. Ecosystem service value for the common reed wetlands in the Liaohe Delta, Northeast China. *Open J. Ecol.* 2016, *6*, 129–137. [CrossRef]
- 124. Ghermandi, A. Integrating social media analysis and revealed preference methods to value the recreation services of ecologically engineered wetlands. *Ecosyst. Serv.* 2018, *31*, 351–357. [CrossRef]
- 125. Vidal Gimenez, F.; Ruiz Mas, C. The Valuation of Recreational Use of Wetlands and the Impact of the Economic Crisis. *Int. J. Environ. Res. Public Health* **2020**, *17*, 3228. [CrossRef]
- 126. Musamba, E.B.; Boon, E.K.; Ngaga, Y.M.; Giliba, R.A.; Dumulinyi, T. The recreational value of Wetlands: Activities, socioeconomic activities and consumers' surplus around Lake Victoria in Musoma Municipality, Tanzania. J. Hum. Ecol. 2012, 37, 85–92. [CrossRef]
- 127. Sinclair, M.; Ghermandi, A.; Sheela, A.M. A crowdsourced valuation of recreational ecosystem services using social media data: An application to a tropical wetland in India. *Sci. Total Environ.* **2018**, *642*, 356–365. [CrossRef]
- 128. Pueyo-Ros, J.; Garcia, X.; Ribas, A.; Fraguell, R.M. Ecological restoration of a coastal wetland at a mass tourism destination. Will the recreational value increase or decrease? *Ecol. Econ.* **2018**, *148*, 1–14. [CrossRef]
- 129. Lankia, T.; Neuvonen, M.; Pouta, E. Effects of water quality changes on the recreation benefits of swimming in Finland: Combined travel cost and contingent behavior model. *Water Resour. Econ.* **2019**, *25*, 2–12. [CrossRef]
- 130. Vesterinen, J.; Pouta, E.; Huhtala, A.; Neuvonen, M. Impacts of changes in water quality on recreation behavior and benefits in Finland. *J. Environ. Manag.* 2010, *91*, 984–994. [CrossRef]
- Lankia, T.; Neuvonen, M.; Pouta, E.; Sievänen, T.; Torvelainen, J. Outdoor recreation in ecosystem service accounting: Pilot accounts from Finland. *Scand. J. For. Res.* 2020, 35, 186–197. [CrossRef]
- 132. Aazami, M.; Shanazi, K. Tourism wetlands and rural sustainable livelihood: The case from Iran. *J. Outdoor Recreat. Tour.* **2020**, 30, 100284. [CrossRef]
- 133. Park, E.; Lee, S.; Peters, D.J. Iowa wetlands outdoor recreation visitors' decision-making process: An extended model of goal-directed behavior. J. Outdoor Recreat. Tour. 2017, 17, 64–76.
- 134. Bryan, B.A.; Raymond, C.M.; Crossman, N.D.; King, D. Comparing spatially explicit ecological and social values for natural areas to identify effective conservation strategies. *Conserv. Biol.* **2011**, *25*, 172–181. [CrossRef] [PubMed]
- 135. Pascual, U.; Balvanera, P.; Díaz, S.; Pataki, G.; Roth, E.; Stenseke, M.; Watson, R.T.; Dessane, E.B.; Islar, M.; Kelemen, E.; et al. Valuing nature's contributions to people: The IPBES approach. *Curr. Opin. Environ. Sustain.* **2017**, *26*, 7–16. [CrossRef]
- 136. Pullin, A.S.; Sutherland, W.; Gardner, T.; Kapos, V.; Fa, J.E. Conservation priorities: Identifying need, taking action and evaluating success. *Key Top. Conserv. Biol.* **2013**, *2*, 3–22.
- 137. Kennedy, C.M.; Oakleaf, J.R.; Theobald, D.M.; Baruch-Mordo, S.; Kiesecker, J. Managing the middle: A shift in conservation priorities based on the global human modification gradient. *Glob. Chang. Biol.* **2019**, *25*, 811–826. [CrossRef] [PubMed]
- 138. Westerberg, V.H.; Lifran, R.; Olsen, S.B. To restore or not? A valuation of social and ecological functions of the Marais des Baux wetland in Southern France. *Ecol. Econ.* **2010**, *69*, 2383–2393. [CrossRef]
- 139. Berkes, F. Evolution of co-management: Role of knowledge generation, bridging organizations and social learning. *J. Environ. Manag.* **2009**, *90*, 1692–1702. [CrossRef]
- 140. Sterner, T. *Policy Instruments for Environmental and Natural Resource Management;* Routledge, 2010. Available online: https://www.routledge.com/Policy-Instruments-for-Environmental-and-Natural-Resource-Management/Sterner-Coria/p/ book/9781617260988 (accessed on 8 November 2021).
- Cottet, M.; Piégay, H.; Bornette, G. Does human perception of wetland aesthetics and healthiness relate to ecological functioning? *J. Environ. Manag.* 2013, 128, 1012–1022. [CrossRef] [PubMed]
- Stephenson, J. The Cultural Values Model: An integrated approach to values in landscapes. *Landsc. Urban Plan.* 2008, 84, 127–139.
 [CrossRef]

- 143. Scott, M.; Lennon, M.; Haase, D.; Kazmierczak, A.; Clabby, G.; Beatley, T. Nature-based solutions for the contemporary city/Renaturing the city/Reflections on urban landscapes, ecosystems services and nature-based solutions in cities/Multifunctional green infrastructure and climate change adaptation: Brownfield greening as an adaptation strategy for vulnerable communities?/Delivering green infrastructure through planning: Insights from practice in Fingal, Ireland/Planning for biophilic cities: From theory to practice. *Plan. Theory Pract.* 2016, *17*, 267–300.
- 144. McCormick, A.; Fisher, K.; Brierley, G. Quantitative assessment of the relationships among ecological, morphological and aesthetic values in a river rehabilitation initiative. *J. Environ. Manag.* **2015**, *153*, 60–67. [CrossRef]
- 145. Brown, G.; Raymond, C. The relationship between place attachment and landscape values: Toward mapping place attachment. *Appl. Geogr.* 2007, 27, 89–111. [CrossRef]
- 146. Gobster, P.H.; Nassauer, J.I.; Daniel, T.C.; Fry, G. The shared landscape: What does aesthetics have to do with ecology? *Landsc. Ecol.* **2007**, *22*, 959–972. [CrossRef]
- 147. Rojas, O.; Zamorano, M.; Saez, K.; Rojas, C.; Vega, C.; Arriagada, L.; Basnou, C. Social perception of ecosystem services in a coastal wetland post-earthquake: A case study in Chile. *Sustainability* **2017**, *9*, 1983. [CrossRef]
- Do, Y.; Kim, J.Y. An assessment of the aesthetic value of protected wetlands based on a photo content and its metadata. *Ecol. Eng.* 2020, 150, 105816. [CrossRef]
- 149. Sonti, N.F.; Campbell, L.K.; Svendsen, E.S.; Johnson, M.L.; Auyeung, D.N. Fear and fascination: Use and perceptions of New York City's forests, wetlands, and landscaped park areas. *Urban For. Urban Green.* **2020**, *49*, 126601. [CrossRef]
- Dobbie, M.F. Public aesthetic preferences to inform sustainable wetland management in Victoria, Australia. Landsc. Urban Plan. 2013, 120, 178–189. [CrossRef]
- Bouahim, S.; Rhazi, L.; Ernoul, L.; Mathevet, R.; Amami, B.; Er-Riyahi, S.; Muller, S.D.; Grillas, P. Combining vulnerability analysis and perceptions of ecosystem services in sensitive landscapes: A case from western Moroccan temporary wetlands. *J. Nat. Conserv.* 2015, 27, 1–9. [CrossRef]
- 152. Torres-Lima, P.; Conway-Gómez, K.; Buentello-Sánchez, R. Socio-environmental perception of an urban wetland and sustainability scenarios: A case study in Mexico City. *Wetlands* **2018**, *38*, 169–181. [CrossRef]
- 153. Manuel, P.M. Cultural perceptions of small urban wetlands: Cases from the Halifax regional municipality, Nova Scotia, Canada. *Wetlands* 2003, *23*, 921–940. [CrossRef]
- 154. Pedersen, E.; Weisner, S.E.; Johansson, M. Wetland areas' direct contributions to residents' well-being entitle them to high cultural ecosystem values. *Sci. Total Environ.* **2019**, *646*, 1315–1326. [CrossRef]
- 155. Siles, G.; Voirin, Y.; Bénié, G.B. Open-source based geo-platform to support management of wetlands and biodiversity in Quebec. *Ecol. Inform.* **2018**, *43*, 84–95. [CrossRef]
- 156. Cohen-Shacham, E.; Walters, G.; Janzen, C.; Maginnis, S. *Nature-Based Solutions to Address Global Societal Challenges*; IUCN: Gland, Switzerland, 2016; Volume 97.
- 157. Buijs, A.; Hansen, R.; Van der Jagt, S.; Ambrose-Oji, B.; Elands, B.; Rall, E.L.; Mattijssen, T.; Pauleit, S.; Runhaar, H.; Olafsson, A.S.; et al. Mosaic governance for urban green infrastructure: Upscaling active citizenship from a local government perspective. *Urban For. Urban Green.* **2019**, *40*, 53–62. [CrossRef]